# Analysts' Strategic Use of Accrual Components

Ivana Raonic

**Cass Business School, London** 

i.raonic@city.ac.uk

## Ali Sahin

**Cass Business School, London** 

March, 2015

Preliminary and incomplete.

#### Abstract

In this paper we examine analysts' strategic use of information in accruals' components. Specifically, we investigate whether analysts' optimism with respect to operating accruals triggers incentives to strategically infuse financial accruals with pessimism to offset the overoptimistic error. We extend the work by Bradshaw et al. (2001) and Drake and Myers (2011) by using total instead of working capital accruals and by breaking total accruals into components with varying degrees of reliability using the approach from Richardson et al. (2005). Our analysis reveals that the negative relation between operating accrual components and forecast errors, i.e., analysts' optimism is a function of a reliability level of a particular component, rather than its magnitude. The results from our main tests suggest that operating accrual-related optimism is offset by financial accrual-related pessimism while the forecast errors associated with total accruals are statistically insignificant from zero. Our findings also indicate that the relations between forecast errors and accrual components are moderated by firm specific unconditional and conditional reporting conservatism. We find that the operating accrual-related optimism further increases and financial accrual-related pessimism decreases for firms with relatively higher unconditional conservatism. On the other hand, we find that conditional conservatism plays little role for operating accrual-related optimism while it further increases financial accrual-related pessimism.

# 1. Introduction

In this paper we investigate analysts' strategic use of information in accruals' components. A number of prior studies examines the relation between analysts' forecasts of earnings and accruals. For example, Bradshaw et al. (2001) investigate whether analysts' forecasts incorporate predictable earnings reversals associated with working capital accruals and document that analyst are over-optimistic about future prospects of firms with high working capital accruals. Similar evidence is provided by a body of research (e.g., Collins et al., 2003, Elgers et al. 2003, Hanlon, 2005, Mashruwala et al. 2006 and Thomas and Zhang, 2002) that builds on Bradshaw et al. (2001) and interprets the findings as analysts' lack of necessary sophistication to understand the implication of high accruals for future earnings. On the other hand, there is a large literature which argues that analysts behave strategically and issue optimistic forecasts due to incentives to curry favour with firm managers (Francis and Philbrick 1993, Richardson, Teoh and Wysocki, 2004, Zhou, 2012). A related set of studies finds that analysts issue more optimistic forecasts for firms whose earnings are more difficult to predict (Das et al., 1998, Ke and Yu, 2006), and that the difficulty of forecasting earnings interacts with analysts' incentives to be optimistic which in turn, results in optimistically biased forecast errors (Bradshaw et al., 2014). Our study is related to each of these literatures and its objective is to investigate whether the varying levels of reliability of accrual components explain analysts' strategic behaviour. We start from the well documented predisposition that analysts have incentives to issue optimistic forecasts and will do so when they hold favourable views of the firms they follow. At the same time, analysts' objective is to provide the most accurate forecasts by minimizing the mean absolute forecast error since forecast accuracy is one the most important indicators of analyst forecast performance (Gu and Wu, 2003). We assume that, in the process of producing earnings forecasts, analysts distinguish between operating and financing accruals and give consideration to the relative reliability and predictability of their different components. We employ a comprehensive categorization of accruals provided by Richardson et al. (2005) in which each accrual category is rated according to its persistence and reliability. Richardson et al. (2005) find that less reliable accruals lead to lower earnings persistence, and argue that a narrow definition of accruals focused on working capital accruals ignores a variety of non-current operating and financial assets and liabilities which results in noisy measures of both accruals and cash flows. In contrast, research that studies relations between analysts' earnings forecasts and accruals (e.g., Bradshaw et al., 2001, Drake and Myers, 2011) focuses on working capital

accruals only, based on an argument that working capital accruals do a better job, relative to total accruals, of capturing the accruals that lead to earnings reversals that are unanticipated by investors. For example, Bradshaw et al. (2001) argue that total accruals include a number of special items such as accruals associated with restructuring and asset impairments, which tend to be non-recurring items on the income statement and thus more likely to be anticipated by investors. However, a more recent evidence in Doyle et al. (2003) suggests that such "special" accruals (e.g., estimated severance costs in restructuring, equity method losses, etc.) omitted from the conventional accrual definition (Sloan, 1996) are far from nonrecurring or unimportant for anticipating future stock returns. In fact, the findings documented in Doyle et al., (2003) show that firms with relatively large omissions of such items in their definitions of pro forma earnings suffer relatively lower stock returns over the next three years and that the investors do not fully appreciate the predictive power of such "special" accruals. We draw on the findings by Doyle et al. (2003) and give consideration to both current and non-current operating accruals in our analysis of analysts' behaviour with respect to accruals. In addition to operating accruals, Richardson et al.'s (2005) definition includes accruals related to noncash financial assets (e.g., investments in long-term bonds) and liabilities (e.g., long-term debt) which are generally measured with high reliability, which form part of earnings (e.g., accrued interest revenue and expense), and which we take into account as an important factor that helps explain analysts' strategic use of accruals.

We argue that inherent properties of operating accruals provide scope to incentive-driven optimism in earnings forecasts. Namely, operating accruals are related to core business activities, and analysts tend to select and recommend firms when they are optimistic about firms' future business prospects. In addition, operating accruals include a variety of items, which contain subjective and unreliable estimates (e.g., allowances for bad debt, cost allocations of inventory, capitalized internally generated intangibles, etc.), that also give rise to unintentional optimistic errors because analysts face greater difficulty in forecasting operating accruals. Hence, with respect to operating accruals we agree with the argument that the combined effect of incentives to be optimistic and forecast difficulty results in optimistic errors (Bradshaw et al., 2014) and we extend this argument by drawing a link between the notion of forecast difficulty and the degree of accrual reliability. However, we find that this argument addresses only one side of the coin. Analysts are likely to be aware of the probability of making optimistic errors with respect to operating accruals, and at the same time they are motivated to arrive at the most accurate forecasts given that the accuracy is the

key indicator of their performance. Analysts are also likely to be aware of the high degree of reliability of financial accruals, which as a result associates them with smaller unintentional forecast errors, and which makes them a potential useful channel trough which analysts might manage forecasts to achieve zero or small forecast error. Therefore, we predict that analysts' optimism with respect to operating accruals triggers incentives to strategically infuse financial accruals with pessimism to offset the overoptimistic error. In order words, we argue that analysts have different incentives in using operating versus financial accruals forecasts as instruments to arrive at the desired forecast of earnings.

The empirical analysis begins by replicating the results in Richardson et al. (2005) to measure the degree of reliability for individual accrual components. Next, we employ the analysts forecast error tests from Bradshaw et al. (2001) which we amend by using total instead of working capital accruals and extend by breaking total accruals into components with varying degrees of reliability. Consistent with prior literature (Bradshaw et al., 2001, Drake and Myers, 2011) we find a negative association between analysts forecast errors and working capital accruals, which is indicative of analysts' optimism. We also find analysts optimism with respect to non-current operating accruals. Moreover, we find that the strength of the negative association between forecast errors and individual operating accrual components increases as the degree of reliability of the particular accrual component decreases. In other words, analysts' forecasts optimism with respect to operating accruals appears to be driven by the level of reliability. Our main empirical prediction is tested by examining whether the forecast optimism related to operating accruals is offset by forecast pessimism with respect to financial accruals. As predicted, we find that forecast errors exhibit financial accrual-related pessimism simultaneously with operating accrual-related optimism, while the errors associated with total accruals are statistically insignificant from zero. Finally, we investigate whether observed relations between forecast errors and accrual components are moderated by firm specific reporting conservatism level. We use two measures of conservatism: unconditional measured by hidden reserves as in Penman and Zhang (2002) and conditional measured by Khan and Watts's (2009) C\_score. We find that operating accrual-related optimism further increases and financial accrual-related pessimism decreases for firms with relatively higher unconditional conservatism (e.g., reflected in unrecorded goodwill) which is expected as analysts are more optimistic in general about firms that offer relatively better future prospects. On the other hand, we find that conditional conservatism plays little role for

operating accrual-related optimism while it further increases financial accrual-related pessimism.

The sensitivity analysis reports analysts' pessimism with respect to cash flows which are the most persistent and reliable among all earnings components. We interpret the results as an indication of analysts' strategic behaviour which is reflected in their differential treatment of particular earnings components conditional on the degree of reliability of the individual component. Our results remain robust after controlling for specific firm characteristics that are considered relevant for earnings forecasts, and after employing a bank of sensitivity analysis tests. Overall, these findings suggest that analysts understand information embedded in accruals and use it strategically in their earnings forecasts.

Our paper makes the following contributions to the existing knowledge. First, we build on the findings by Bradshaw et al. (2001), by showing that analysts' forecasts optimism extends to non-current operating accruals and that the reliability of individual accrual components is important to explain the variation of operating accrual-related optimism. Second, our results reveal that analysts forecast errors are mainly driven by accrual *reliability* rather than by the magnitude<sup>1</sup>, i.e., we show that high magnitude of an accrual component does not necessarily lead to larger forecast errors (and larger negative stock returns) if the reliability of the component is high. This finding provides further support to Richardson et al. (2005) who link the persistency of accruals to the reliability concept of accounting, and document that as the persistency decreases investors suffer bigger losses. Third, we demonstrate that operating accrual-related optimism is mitigated by financial accrual-pessimism which results in more accurate earnings forecasts in the aggregate, and which is indicative of analysts understanding properties and implications of various accrual components and acting strategically to minimize the forecast error.

Finally, we demonstrate that analysts give consideration to conservative accounting that shapes the manner in which firms' report accruals, and show that unconditional and conditional conservative accounting practices have different forecasting implications where unconditional conservatism leads to greater (smaller) optimism (pessimism) while conditional conservatism leads to greater financial accrual-related pessimism.

<sup>&</sup>lt;sup>1</sup> Earnings fixation hypothesis suggests that investors fail to anticipate that high accruals do not persist in future which lead to lower subsequent earnings, and negative stock returns.

The remainder of the paper is organised as follow. The next section discusses prior literature and sets empirical predictions. Section 3 describes the data. Section 4 explains research design and presents the results. Section 5 reports sensitivity analyses and Section 6 concludes.

## 2. Hypotheses

Prior research on the analysts' understanding of accruals reports accrual-related optimism and generally explains this as analysts' lack of sophistication and experience to fully appreciate predictable earnings reversals associated with high accruals (Bradshaw et al., 2001, Drake and Myers, 2011). To an extent these results are consistent with the arguments in numerous accounting studies (e.g., Sloan, 1996) which claim that investors are optimistic about the firms with high accruals since they do not appear to anticipate future earnings decreases associated with high accruals. On the other hand, more recent research indicates that it might not be the accrual *magnitude* that drives investors' optimism. For example, Kraft et al. (2006) provide evidence that investors are optimistic about both high and low accruals, Konstantinidi et al. (2012) document that investors rationally distinguish between accruals and cash flows, and respond differently to their surprises. In the present study, we give consideration to these arguments and propose that analysts are likely to differentiate across a range of accruals with different properties, and that analysts' behaviour is likely to be influenced by a set of accruals' attributes other than their magnitude. We also note that the research on analysts' understanding of accruals (e.g., Drake and Myers, 2011), focuses on working capital accruals arguing that they better capture, relative to total accruals, items that lead to earnings reversals unanticipated by investors. Total accruals include a number of noncurrent items such as accruals associated with asset impairments and equity method losses, which Bradshaw et al. (2001) consider as typically non-recurring, flagged in the balance sheet, and hence likely to be anticipated by investors. On the other hand, evidence in Doyle et al. (2003) shows that these categories of accruals are far from non-recurring or unimportant for predicting future returns. They document that firms with relatively large omissions of such items from their pro forma earnings exhibit relatively lower future stock returns, which is indicative of investors not fully appreciating the predictive power of these accruals. Along the similar lines, Richardson et al. (2005) argue that the traditional accrual definition based

on the working capital (Healy, 1985) is too narrow and document that omitting a large part of accruals such as non-current operating and financial accruals causes significant information losses. By using a comprehensive categorisation of accruals, they rate each accrual category according to the reliability of the underlying accrual components and demonstrate that less reliable accruals lead to lower earnings persistence. Their findings offer two major implications that we consider worth analysing in an attempt to explain analysts' behaviour. First, if analysts use all relevant information to produce forecasts, then, they are likely to take into account all accrual components rather than working capital accruals only. Second, each accrual component has distinct reliability characteristics determined mainly by the underlying component's nature. For example, inventories are subject to measurement errors more than bank loans, and hence less reliable regardless of their magnitude. Also, analysts are more likely to make unintentional errors when forecasting less reliable accruals, since they face greater difficulty in forecasting them. Evidence in Das et al. (1998) and Ke and Yu (2006) suggest that analysts issue more optimistic forecasts of earnings for firms whose earnings are more difficult to predict. Hence, given these arguments, we include all accrual information in our tests, and investigate whether analysts' forecasts are influenced by the varying degree of reliability of different accrual parts.

We build our argument on the well documented predisposition that analysts have incentives to issue optimistic forecasts and will do so when they hold favourable views of the firms' performance they follow (see for example Bradshaw (2011), Francis and Philbrick (1993), McNichols and O'Brian (1997), Michaely and Womack (1999), Hong and Kubic (2003)). Firms performance and future prospects are best captured by their operating activities (Feltham and Ohlson (1995), Nissim and Penman (2003), Penman et al. (2006)) which are core to the business, and which are strongly linked to firm's goodwill. We argue that analysts distinguish across accruals that reflect a range of operating and financial activities and have incentives to be optimistic with respect to operating accruals. Given that operating accruals include a number of components with a varying degree of reliability and that less reliable operating accruals are likely to be inherently more difficult to forecast, with respect to operating accruals, we agree with the argument that forecasting difficulty interacts with analysts incentives to be optimistic and results in optimistic forecast errors (Bradshaw et al., 2014). For example, operating accruals contain subjective estimates such as allowance for bad debt, cost allocation of inventory, capitalized internally generated software and other intangibles, which in forecasting open a scope for unintentional optimistic errors. The possibility of such errors is likely to be greater for less reliable accrual estimates, so we predict that analysts' accrual-related optimistic errors will increase as the degree of reliability of a particular accrual component decreases.

We recognize that analysts may have different incentives in forecasting operating versus financial accruals. Investors and analysts may not attribute the same degree of importance to financial accruals compared to operating accruals<sup>2</sup>. Also, by their nature, financial accruals (such as accrued interest revenue and expenses) are less subject to managerial discretion and estimation errors, and therefore more reliable compared to operating accruals. Analysts are aware of the high degree of reliability of financial accruals and they are likely to be less concerned about the subsequent realisations of financing compared to operating accruals. Our key hypothesis is based on the assumption that analysts select earnings components which are less subject to unintentional errors and/or discretionary biases for strategic purposes because the desired outcomes of their strategies are then likely to be more accurate. Therefore, we hypothesize that analysts use financial accruals in a strategic manner by infusing them with pessimism in order to offset the operating accrual-related optimism<sup>3</sup>.

Consider future actual earnings X at time t+1, which consists of three components:

$$X = A^o + A^f + CF \tag{2}$$

Where  $A^o$  and  $A^f$  denote operating and financial accruals respectively, while *CF* denotes cash flow. An analyst estimates a (draft) forecast of *X* between the announcements of time *t* and *t*+1 actual earnings

$$FX = FA^o + FA^f + FCF \tag{3}$$

Next, we define forecasts of accruals and cash flows as:

<sup>&</sup>lt;sup>2</sup> Investors and analysts tend to pay attention to price multiples of accounting numbers that reflect operating activities such as P/EBIT, P/EBITDA and P/Sales (see for example Penman, (2007), Kim et al. (2012)).

<sup>&</sup>lt;sup>3</sup> Matsumoto (2002) and Burgstahler and Eames (2006) provide evidence of the downward management of analysts' forecasts in order to achieve zero or small positive earnings surprises. While this evidence suggests analysts' strategic use of earnings forecasts, the actual channels (in terms of earnings components) through which such strategies are carried out, remain unexplored.

$$FA^o = A^o + e^o \tag{4}$$

$$FA^f = A^f + e^f \tag{5}$$

$$FCF = CF + e^{cf} \tag{6}$$

The forecast with the following properties is considered as the most accurate with regards to accruals (we do not make any assumption about *CF* at this stage):

$$e^o = e^f = 0$$
, i.e., accrual related error terms are zero

$$e^{o} + e^{f} = 0$$
 or min., i.e., accrual related error terms offset each other

In our framework, the second option suggests a strategic treatment of accrual information by analysts where  $e^o > 0$  reflects operating accrual-related optimistic errors, and  $e^f < 0$  reflects strategic pessimistic errors.

### 2.1 Strategic forecasting and financial reporting

Analysts' understanding and use of accrual components is likely to be influenced by firms' financial reporting characteristics. One of the key attributes of financial reporting is accounting conservatism. The literature typically distinguishes between two forms of conservatism: conditional and unconditional. Conditional conservatism (Basu, 1997) is reflected in a greater sensitivity of contemporaneous earnings with regard to economic losses relative to gains implied by negative and positive stock value changes, and as a result, book values are written down when the news are bad, but not written up to the same extent, when the news are good. Under unconditional conservatism, accounting generates pervasive bias regardless of the news (Beaver & Ryan, 2005), and gives rise to hidden reserves (i.e., unrecorded goodwill) by means of an immediate expensing of R&D, creation of a LIFO reserve, etc.

Consistent with the argument presented above that analysts have incentives to be optimistic when they hold favourable views of the firms they follow, we investigate whether firms with relatively larger unconditional conservatism (i.e., unrecorded goodwill) are favoured relatively more by analysts. Since a firm's performance and future prospects are best captured by its operating activities, and since goodwill is mostly attributable to firm's operations, we test whether operating accrual-related optimism is greater for more unconditionally conservative firms. We measure unconditional conservatism by hidden reserves introduced by Penman & Zhang (2002).

Prior research also indicates that high unconditional conservatism leads to a less conditional conservatism (e.g., Beaver & Ryan, 2005; Roychowdhury & Watts, 2007), suggesting that each form of conservatism may reflect different aspects of variations in forecast errors. For example, Helbok & Walker (2004) document that analysts forecast bias is greater for firms with larger conditional conservatism, and Konstantinidi et al. (2012) document that the application of conditional conservatism causes asymmetry in the mean reversion of accruals, because immediately recognised losses are transitory shocks to the earnings process, which leads to a lower persistency of accruals. In the context of the present study, this evidence could imply that relatively less persistent accruals due to relatively larger conditional conservatism are likely to cause bigger forecast errors. To the extent that analysts are able to unravel future implications of firm's conditional conservatism and accordingly adjust earnings forecasts, considering that the asymmetric loss recognition is usually reflected in accounting numbers that capture operating activities, we expect operating accrual-related optimism to be smaller for firms that are consistently conservative. At the same time, timely loss recognition and the related lower persistency of operating accruals increase the possibility of unintentional forecast errors  $(e^{0} > 0)$ . In order to minimize the overall unintentional error, we expect analyst to strategically reduce their forecasts with respect to financial accruals leading to relatively larger pessimistic errors  $e^{f} < 0$  for firms with larger conditional conservatism. We measure firm specific conditional conservatism by the C-Score construct introduced by Khan & Watts (2009) and derived from the Basu (1997) model.

#### 3. Data

The sample used for the tests consists of non-financial US firms for the period between 1976 and 2013. Financial statement data is obtained from Compustat annual database. Analysts forecast data is from the IBES summary statistics file and stock returns data are from CRSP

daily files. To decompose earnings into components we rely on the accrual definition from Richardson et al.  $(2005)^4$ :

$$Total\ accruals = \Delta WC + \Delta NCO + \Delta FIN \tag{7}$$

where  $\Delta WC$  is the change in non-cash working capital,  $\Delta NCO$  is the change in net noncurrent operating assets and  $\Delta FIN$  is the change in net financial assets;

and rewrite equation (2) as:

$$X = \Delta WC + \Delta NCO + \Delta FIN + CF \tag{8}$$

where  $\Delta WC + \Delta NCO = A^o$ , and  $\Delta FIN = A^f$  from equation (2).

Total accruals (TACC) can be further decomposed into their underlying components:

$$TACC_{t} = \underbrace{\Delta COA_{t} - \Delta COL_{t}}_{\Delta WC} + \underbrace{\Delta NCOA_{t} - \Delta NCOL_{t}}_{\Delta NCO} + \underbrace{\Delta STI_{t} + \Delta LTI_{t} - \Delta FINL_{t}}_{\Delta FIN}$$
(9)

All the variables are defined in Table A in the Appendix. Following Richardson et al. (2005), the missing data on short term debt, investment and advances, and long term debt are set to zero, while all other missing observations are eliminated. All earnings, accruals and cash flows variables are winsorised to +1 and -1 and deflated by average assets to eliminate extreme observations. Analyst forecast errors, conservatism proxies (*Hidden\_reserves* and *C\_Score*), returns, and continuous control variables are winsorised to 1% and 99%. Our final sample contains 142,821 firm-year observations for the accrual reliability tests, and it ranges between 45,149 and 48,142 firm years for the forecast error regressions.

Table 1 Panel A sets out descriptive statistics for earnings (*ROA*), total (*TACC*), operating (*OPAC*), and financial ( $\Delta FIN$ ) accruals, working capital ( $\Delta WC$ ) and non-current ( $\Delta NCO$ ) accrual components. It also includes descriptives for conservatism proxies, *Hidden\_reserves* and *C\_Score*. Mean *TACC* is 0.051 or roughly 5% of total assets. Means of working capital accruals ( $\Delta WC$ ) and non-current operating accruals ( $\Delta NCO$ ) are positive while while mean finacial accruals ( $\Delta FIN$ ) is negative, which is indicative of an average firm increasing its non-current operations and financing this increase by net debt. Panel B reports pairwise

<sup>&</sup>lt;sup>4</sup>Richardson et al. (2005) define accrual-based earnings through the definition of net assets: Accrual earnings =  $\Delta$ Assets –  $\Delta$ Liabilities + Net cash distribution. Given that Accruals = Accrual earnings – Cash earnings, and that Cash earnings =  $\Delta$ Cash + Net cash distribution, they define Accruals =  $\Delta$ Assets – $\Delta$ Liabilities –  $\Delta$ Cash.

correlations, and reveals that all accrual components are positively correlated with *ROA*, with  $\Delta WC$  having the highest correlation. The positive correlation between  $\Delta WC$  and  $\Delta NCO$  suggests that they grow together. Both  $\Delta WC$  and  $\Delta NCO$  are negatively correlated with  $\Delta FIN$ , which is in line with the suggestion that growth in operating activities is largely financed by debt.

Table 2 presents the descriptive statistics and pairwise correlations for the extended accrual decomposition. Panel A shows that mean values of all accrual components are positive and that  $\Delta NCOA$  (change in non-current assets) have the highest mean (0.055) while  $\Delta LTI$  (change in long term investments), and  $\Delta COA$  (change in current operating assets) have the lowest means (0.002 and 0.004, respectively) suggesting that non-current operating accruals constitute the major part of accruals. Looking at the standard deviations of accruals, the results suggest that much of the variation in working capital accruals is attributed to  $\Delta COA$ . Similar pattern is found with respect to non-current operating accruals, which implies that the asset side of operating of accruals is more likely to be subject to measurement errors. In contrast, much of the variation in  $\Delta FIN$  can be attributable to financial liability,  $\Delta FINL$ . These observations suggest that the variation in operating accruals are driven by liabilities.

Panel B of Table 2 reports pairwise correlations and shows strong correlation among accrual components. In particular, there is positive correlation between  $\triangle COA$  and  $\triangle COL$  suggesting that a growing (shrinking) business generally results in an increase (decrease) in both current operating assets and liabilities. There is also a positive correlation between  $\triangle COA$  and  $\triangle FINL$  suggesting that current operations are not only funded by operating liabilities, but also by financial debt. Moreover,  $\triangle NCOA$  is positively correlated with all liability accruals<sup>5</sup>.

# 4. Empirical analysis

Following previous work in the area, notably Bradshaw et al. (2001) and Richardson et al. (2005) an association study is conducted in order to assess the relative degree of reliability of

<sup>&</sup>lt;sup>5</sup> Note that the liability component of accruals is substracted from the asset component to arrive at net accruals. Hence, a positive relation between an asset and a liability implies that they are likely to offset each other's effect on net accruals.

each accrual component. As a first step, to check the relative persistance of total accruals and cash flows, we estimate the following model from Richardson et al. (2005):

$$ROA_{t+1} = \gamma_0 + \gamma_1 (ROA_t - TACC_t) + \gamma_2 TACC_t + \vartheta_{t+1}$$
(10)

Where *ROA* represents earnings scaled by total assets, (*ROA* – *TACC*) cash flows, and *TACC* total accruals. As accrual component is expected to be less persistent than the cash flow component of earnings, the prediction is:  $(\gamma_2 - \gamma_1) < 0$ . Next, to measure the relative persistence of each accrual component model (10) is expanded as follows:

$$ROA_{t+1} = \gamma_0 + \gamma_1 (ROA_t - \Delta WC_t - \Delta NCO_t - \Delta FIN_t)$$
$$+ \gamma_2 \Delta WC_t + \gamma_3 \Delta NCO_t + \gamma_4 \Delta FIN_t + \vartheta_{it+1} \quad (10a)$$

In equation (10a) the coefficient  $\gamma_1$  measures the persistence of the cash flow component, while( $\gamma_2 - \gamma_1$ ), ( $\gamma_3 - \gamma_1$ ), and ( $\gamma_4 - \gamma_1$ ) measure the persistence of  $\Delta WC$ ,  $\Delta NCO$ , and  $\Delta FIN$ relative to cash flow component, respectively. To directly estimate  $\gamma_k$  (where k = 2, 3, and 4) relative to  $\gamma_1$ , we rewrite equation (10a) and fit the following model:

$$ROA_{t+1} = \rho_0 + \rho_1 ROA_t + \rho_2 \Delta WC_t + \rho_3 \Delta NCO_t + \rho_4 \Delta FIN_t + \vartheta_{it+1}$$
(10b)

We conduct our regression analyses using ordinary least squares (OLS) and cluster standard errors by firm and year following Petersen  $(2009)^6$ . In the robustness tests, we also employ the Fama & MacBeth (1973) procedure.<sup>7</sup>

Results are reported in Table 3. Results in panel A confirm prior findings that earnings are slowly mean reverting, and that the mean reversion of accruals are quicker than cash flows, i.e., accruals are less persistent than cash flows (the persistence coefficient on *ROA* is 0.80

<sup>&</sup>lt;sup>6</sup> Petersen (2009) method deals with the potential time and firm effects that can be present in panel data sets. Firm effect (or time series dependence) means the residuals may be correlated across years for a given firm, and time effect (or cross sectional dependence) means the residuals of a given year may be correlated across different firms. He documents that in the presence of both firm and time effects, clustering the standard errors by two dimensions simultaneously yields the *unbiased estimates* as long as there are sufficient number of clusters. For more information about clustering the standard errors, see Petersen (2009), and to obtain the programming advice visit

http://www.kellogg.northwestern.edu/faculty/petersen/htm/papers/se/se\_programming.htm

<sup>&</sup>lt;sup>7</sup> Fama & McBeth (1973 procedure estimates annual cross sectional regressions and report the time series average of the resulting coefficients.

while the coefficient on TACC is negative, and both are significant at 1% level). Panels B and C report persistence results for the accrual decomposition to  $\triangle OPAC$  and  $\triangle FIN$ , and to  $\triangle WC$ ,  $\Delta NCO$ , and  $\Delta FIN$ , respectively. Consistent with the Richardson et al. (2005) and as predicted the results indicate that operating accruals are less persistent than financial accruals. For example, Panel C shows that the magnitude of the persistence coefficients ranges from -0.137 for  $\Delta WC$  to -0.045 for  $\Delta FIN$  (all with p-values<0.01). Finally, Panel D sets out the results for the extended accrual decomposition and reports the relative persistence of each accrual component. The coefficients on  $\triangle COL$  and  $\triangle COA$  are the lowest (-0.177 and -0.132, both significant at 1% level), while the coefficients on  $\triangle COL$  and  $\triangle NCOA$  are -0.097 and -0.077, respectively (both significant at 1%). With regards to financial accruals, a similar pattern is observed: the coefficient ranges from -0.059 for  $\Delta LTI$  (which includes less reliable items such as long term receivables), to -0.034 for  $\Delta STI$  (e.g., marked-to-market short term investments). In summary, the preceding analysis provides two objectives. First, the results confirm the reliability ranks of accruals from Richardson et al. (2005) which range from the least (working capital) to the most (financial) accruals. Second, the findings provide us with the measure of reliability for each accrual components which we use to explain analysts strategic behaviour with respect to accruals below.

We employ Bradshaw et al. (2001) methodology to investigate the relation between analysts forecast errors and individual accrual components. We use IBES consensus median forecast earnings and measure forecast error as the difference between realized earnings in year t+1 and monthly analysts expectation of earnings for the forthcoming year (t+1) immediately after the release of year t earnings. We deflate all forecast errors by stock prices from CRSP at the end of year t. We regresses 12 consecutive months' forecast errors on current period t accrual components as follows:

$$Ferror_{s,it+1} = \beta_0 + \beta_1 \Delta W C_{it} + \beta_2 \Delta N C O_{it} + \beta_3 \Delta F I N_{it} + \varepsilon_{it+1}$$
(11)

Our main empirical prediction is that analysts' optimism with respect to operating accruals triggers incentives to infuse financial accruals with pessimism to offset the overoptimistic error. In equation (11) a negative forecast error indicates that the forecast is optimistic. Therefore, we expect negative coefficients on  $\Delta WC$  and  $\Delta NCO$  and a positive coefficient on  $\Delta FIN$ . The results are reported in Table 4. In line with our predictions, analysts' optimism

prevails in WC accruals; the coefficients on  $\Delta WC$  are negative and significant across 12 months, while the coefficients on  $\triangle NCO$  are not statistically different from zero. Analyst optimism is driven by working capital accruals as predicted that are less persistent, less reliable and more difficult to predict than non-current operating activity accruals. Since current operations are more likely to provide relevant information about firm's daily activities, cash flows, profitability, operating cycle, liquidity, etc., analysts' focus on current operations is expected. Also, the results in Panel A also confirms that the strength of association between forecast errors and accrual components is gradually decreasing as time horizon extends consistent with previous research (e.g. Bradshaw et al., 2001) since the actual earnings outcome become more predictable. To empirically check the offsetting effects between operating-accrual related optimism and financial-accrual related pessimism, we regress forecast errors on total accruals. Results reported in Panel B of Table 4 reveal that, in the aggregate, analysts forecast errors are not associated with total accruals. The relation is statistically and economically zero across all 12 months confirming our prediction. Next, to empirically test the relation between analysts optimism and the degree of reliability of particular accrual components we rerun regression model (11) extending the accrual decomposition of  $\Delta WC$  and  $\Delta NCO$  into their asset and liability parts. The results reported in Table 5 show significant analyst optimism with respect to current operating asset accruals (COA), current operating liability accruals (COL), and also non-current operating liability accruals (NCOL). More importantly, the results reveal a pattern where operating accrualrelated optimism decreases as the degree of persistence of an individual component increases. For example, analysis in Panel D, Table 3 suggests that  $\triangle COL$  is relatively less persistent and reliable (e.g.  $\triangle COL$  is likely to include various short-term provisions whose future implications are relatively more difficult for analyst to predict), than  $\triangle COA$  which is less persistent than  $\Delta NCOL$  while  $\Delta NCOA$  seems to be the most reliable operating accrual component (i.e., long-term operating assets are usually the most straightforward to measure). The relative rankings of persistence coefficients on the accrual components line up closely with their counterparts in Table 5: optimistic errors are the largest for the least persistent components, and they monotonically decrease as the persistence rank of a particular accrual component increases<sup>8</sup>.

<sup>&</sup>lt;sup>8</sup> Untabulated *F-tests* reveal that the coefficients in table 10 are different from each other.

Next, to check whether the findings in main tests remain robust if we incorporate firm specific controls, we re-estimate equation (11) with the extended accrual decomposition controlling for firm size, beta, B/P, leverage, loss dummy, E/P and past returns with industry fixed effects. All the control variables are defined in Table A of the Appendix. The regressions are run for the first month just after the announcement of year *t* earnings. Table 6 reports the results and reveals several findings. First, the explanatory power of regressions increases significantly with controls. All controls are significantly associated with forecast errors, and the signs of coefficients on control variables are generally consistent with prior research except for the signs of past year returns (e.g., Abarbanell, 1991), firm size (e.g., Keskek & Tse, 2013), and beta (e.g., Ljungqvist, Marston, Starks, Wei, and Yan, 2007). Finally, the main coefficients capturing analyst optimism/pessimism remain robust. One exception is that, when all controls are incorporated, financial accrual-related pessimism with respect to  $\Delta FINL$  diminishes. The unreported tests reveal that this drop in pessimism can be explained by relatively high leverage, which in turn is related to relatively higher future earnings and profitability.

#### 4.1 Effects of accounting conservatism

The next set of tests investigates whether relations between forecast errors and accruals components are moderated by firm specific conservatism level. To measure unconditional conservatism we compute *Hidden\_Reserves*<sub>t</sub> =  $R\&Dres_t + ADVres_t + LIFOres_t$  from Penman and Zhang (2002, 2013). The definitions of the variables are provided in Table A of the Appendix. To measure conditional conservatism we estimate the model suggested by Khan and Watts (2009) based on Basu's (1997) measure of asymmetric timeliness of bad relative to good news ( $X_{it} = \beta_1 + \beta_2 D_i + \beta_3 R_{it} + \beta_4 D_i R_{it} + e_{it}$ , where the asymmetric timeliness is measured by the differential coefficient  $\beta_4$ )<sup>9</sup>. Since firm-specific characteristics - size, market to book ratio and leverage – have been found to significantly affect conservatism, Khan & Watts (2009) transform the Basu (1997) model to obtain weights used to aggregate firm characteristics into annual firm-specific measures of timeliness<sup>10</sup>:

$$\begin{aligned} X_t &= \beta_1 + \beta_2 D_i + R_i (\mu_1 + \mu_2 Size_i + \mu_3 M/B_i + \mu_4 lev_i D_i) + D_i R_i (\gamma_1 + \gamma_2 Size_i + \gamma_3 M/B_i + \gamma_4 lev_i) + \delta_1 Size_i + \delta_2 M/B_i + \delta_3 Lev_i + \delta_1 D_i Size_i + \delta_2 D_i M/B_i + \delta_3 D_i Lev_i + e_{it} \end{aligned}$$

<sup>&</sup>lt;sup>9</sup> See Basu (1997) for the model development.

<sup>&</sup>lt;sup>10</sup> See Khand and Watts (2009) for the model's development and the validation of the conservatism score.

Asymmetric timeliness with respect to bad news ( $C\_Score$ ) is a linear function of firmspecific characteristics and calculated for each firm and year:

$$C\_Score \equiv \beta_4 = \gamma_1 + \gamma_2 Size_i + \gamma_3 \frac{M_i}{B_i} + \gamma_4 lev_i$$

The definitions of the variable are in Table A. To distinguish between firms with high versus low unconditional (conditional) conservatism, we group the sample into quintiles based on the magnitude of *Hidden\_reserves (C\_score)* for the first three months of year *t* before the announcement of quarterly earnings and we scale the quintiles to a (0,1) range so that the lowest (highest) quintile is assigned a value of 0 (1). We also independently stratify the sample into quintiles based on the magnitude of accruals in year *t* for each accrual component ( $\Delta WC$ ,  $\Delta NCO$  and  $\Delta FIN$ ). We then extend the main forecast error model (11) including quintile conservatism ranking and its interactions with each of the accrual components ( $\Delta WC$ ,  $\Delta NCO$  and  $\Delta FIN$ ) quintiles, respectively:

$$Ferror_{1,it+1} = \beta_0 + \beta_1 D_i + \sum_{k=1}^3 \delta_k QACC_{it} + \sum_{i=1}^3 \gamma_k D_i * QACC_{it} + \varepsilon_{it+1}$$
(12)

Where QACC is a particular accrual component quintile and D is a conservatism dummy which differentiates between low and high *Hidden\_reserve* (C\_score) firm-years, as explained above. Based on our hypotheses in Section 2.1 the empirical predictions are as follows. With respect to *Hidden\_Reserve* we expect the coefficient  $\gamma_k$  to be significantly negative indicating (i) larger operating accrual-related unconditional conservatism and (ii) a mitigating effect on financial accrual-related pessimism. With respect to C\_Score we expect for firms that are consistently conservative, a significantly positive  $\gamma_k$  which is indicative of relatively smaller operating accrual-related optimism and relatively larger financial accrualrelated pessimism. The results provided in Table 7 provide support for our predictions. For example, for firms with high Hidden\_reserves, the optimism is increasing in working capital (the coefficient on the interaction  $D^*Q \Delta WC$  is -0.021 and statistically significant at the 5% level), while the financial accrual-related pessimism is decreasing (e.g., the average pessimistic error drops from 4% for firms with low hidden reserves to 1.1% for firms with high hidden reserves). On the other hand, we find no evidence of analysts reducing operating accrual-related optimisms for high *C\_score* firms, but they do appear to be incrementally pessimistic with respect to financial accruals (on average the pessimistic error for high *C\_score* firms is 3.6% compared to the 1.6% for low *C\_score* companies).

#### 5. Sensitivity analyses

The empirical analysis so far has found that analyst forecasts are pessimistic with respect to financial accruals which by nature are close to cash flows. As a sensitivity analysis, we now test whether forecast with respect to cash flow component of earnings are also pessimistic, and whether the pessimistic bias is relatively higher compared to financial accruals. This test is conducted by fitting the following models.

$$Ferror_{s,it+1} = \beta_0 + \beta_1 \Delta OPAC_{it} + \beta_2 \Delta FIN_{it} + \beta_3 CF_{it} + \varepsilon_{it+1}$$
(13)

$$Ferror_{s,it+1} = \beta_0 + \beta_1 \Delta LT I_{it} + \beta_2 \Delta FINL_{it} + \beta_3 \Delta ST I_{it} + \beta_4 CF_{it}$$
(14)

In line with the hypotheses explained in Section 2 that analysts are prepared to use more reliable earnings components in a strategic manner in order to offset the operating accrual-related optimisms, we expect positive coefficients on *CF*. We also expect that the coefficient on *CF* is higher in magnitude than coefficients on  $\Delta FIN$ ,  $\Delta LTI$ ,  $\Delta FINL$  and  $\Delta STI$ , respectively, which is indicative of analysts strategically using earnings components that are associated with least chances of unintentional or discretionary errors to. Results reported in Panels A and B in Table 8 are consistent with our predictions. For example, coefficients on *CF* are positive and significant (p < 0.01) across 12 months, and highest relative to all other coefficients from both equations.

Another set of robustness tests investigates whether investors anticipate the implications of operating and financial accruals for future earnings. Prior research (Richardson et al., 2005) provides evidence that investors behave in a naïve manner and fail to anticipate lower persistence of less reliable accrual components which results in negative future stock returns (i.e., in a negative relation between less persistent accruals and future returns). To check the validity of our hypothesis of analysts' strategic use of accruals we examine whether investors behave in a similar manner to analysts. We argue that there is no reason for investors to be optimistic with respect to operating and at the same time strategically pessimistic with respect to financial accruals. Therefore, while we do expect a negative relation between less persistent accruals and future returns. The following model tests these predictions:

$$Ret_{it+1} = \beta_0 + \sum_{k=1}^n \beta_k ACC_{it} + \sum_{j=1}^m \beta_{k+j} X_{it} + \varepsilon_{it+1}$$
(15)

Where X denotes control variables (firm size, market Beta, B/P, E/P and past returns), Ret is the size-adjusted 12 months stock returns for firm i at time t+1 calculated as the buy-andhold returns inclusive of dividends minus the buy-and-hold returns on value-weighted decile portfolios. Return accumulation starts in the fourth month after the fiscal year end, and continues for the next twelve months. Panels A and B of Table 9 report the results which show that market is optimistic with regard to all accrual components (and total accruals in Panel A), and that the least persistent accruals are the most negative and significant. Contrary to the results from forecast errors regressions we find no financial accrual-related pessimism. These findings provide additional support to our argument that analysts do strategically use accrual information. Panel B also shows that the relative order of the coefficients on the accrual components follows a similar pattern found in Richardson et al. (2005) who document that the negative relation between accruals and future stock returns is stronger for less persistent accruals. Furthermore, the coefficients are higher in magnitude compared to the ones from the forecasts error regressions which lends support to the argument that analysts are more sophisticated users of accrual information relative to market investors (Elgers et al., 2001, 2003)<sup>11</sup>.

In further robustness tests we replace actual values of accruals with their decile ranks in forecast error regressions following prior literature (e.g., Abarbanell & Bushee, 1998; Bradshaw et al., 2001; Collins et al., 2003). We group firms annually in 10 portfolios based on the magnitude of accruals in year t, and scale them to a (0,1) range<sup>12</sup> so that lowest (highest) accrual firm-years are assigned a value of 0 (1). The scaling is used to alleviate nonlinearities in the data, and to minimize the effects of measurement errors. The intercept in the decile rank regressions measures the average forecast error for a low accrual firm-year while the coefficient on the particular accrual component rank measures the average incremental forecast errors for a high versus a low accrual firm-year. Results set out in Panels A and B of Table 10 show the same patterns as those observed in Tables 4 and 5. Overall, the findings indicate that it is the accrual persistence rather than their magnitude which drives the forecast errors<sup>13</sup>.

<sup>&</sup>lt;sup>11</sup> Unreported tests which control for regulatory changes such as the enactment of *Regulation Fair Disclosure* and *the Global Analyst Research Settlement* in 2002 reveal that the results remain robust.

<sup>&</sup>lt;sup>13</sup> We also run regressions using Fama & MacBeth (1973) cross sectional regressions and report the time series average of the resulting coefficients. The results remain robust. Tests are also conducted using the data post

# 6. Conclusion

In this paper we examine analysts' strategic use of information in accruals' components. Specifically, we investigate whether analysts' optimism with respect to operating accruals triggers incentives to strategically infuse financial accruals with pessimism to offset the overoptimistic error. We extend the work by Bradshaw et al. (2001) and Drake and Myers (2011) by using total instead of working capital accruals and by breaking total accruals into components with varying degrees of reliability using the approach from Richardson et al. (2005). Our analysis reveals that the negative relation between operating accrual components and forecast errors, i.e., analysts' optimism is a function of a reliability level of a particular component, rather than its magnitude. The results from our main tests suggest that operating accrual-related optimism is offset by financial accrual-related pessimism while the forecast errors associated with total accruals are statistically insignificant from zero. Our findings also indicate that the relations between forecast errors and accrual components are moderated by firm specific unconditional and conditional reporting conservatism. We find that the operating accrual-related optimism further increases and financial accrual-related pessimism decreases for firms with relatively higher unconditional conservatism. On the other hand, we find that conditional conservatism plays little role for operating accrual-related optimism while it further increases financial accrual-related pessimism.

<sup>1993</sup> in order to check whether the main findings alter after the change in IBES's method of calculating earnings (Konstantinidi et al. 2012). The results confirm the initial findings.

# References

- Abarbanell, J. & V. Bernard (1992), "Tests of analysts' overreaction/underreaction to earnings information as an explanation for anomalous stock price behavior," *Journal of Finance*, vol. 47, pp. 1181–1207.
- Abarbanell, J. (1991) "Do analysts' earnings forecasts incorporate information in prior stock price changes?" *Journal of Accounting and Economics* vol.14 (2): 147-65.
- Abarbanell, J., & Bushee, B. (1997), "Fundamental analysis, future earnings, and stock prices," *J. Accounting Research*, vol. 35, pp. 1–24.
- Abarbanell, J., & Bushee, B. (1998) "Abnormal returns to a fundamental analysis strategy". *The Accounting Review* 73 (1), 19–45.
- Ali, A., Klein, A., and Rosenfeld., J. (1992) "Analysts" use of information about permanent and transitory earnings components in forecasting annual EPS". *The Accounting Review*, 67, 183–198.
- Barefield, R. & Comiskey, E. (1975), "The Accuracy of Analysts' Forecasts of Earnings per Share," *J. Business Research*, vol. 3, no. 3, pp. 241–252,
- Barth, Mary & Hutton, A. (2004) "Analyst earnings forecast revisions and the pricing of accruals" *Review of Accounting Studies* vol.9: 59-96.
- Basu, S. &S. Markov (2004) "Loss Function Assumptions in Rational Expectations Tests on Financial Analysts' Earnings Forecasts", J. of Accounting & Economics, 38, 171-203
- Basu, Sudipta (1997) "The Conservatism Principle and the Asymmetric Timeliness of Earnings", *Journal of Accounting & Economics*, Vol. 24, No. 1, 3-37.
- Beaver, W.H., & S. Ryan (2005) "Conditional and unconditional conservatism: concepts and modelling" *Review of Accounting Studies* vol.10, no.2/3, pp.269-309.
- Biddle. G.C. & W.E. Ricks (1988), "Analyst forecast errors and stock price behavior near the earnings announcement dates of LIFO adopters," *J. Accounting Research*, vol. 26, pp. 169–194.
- Bradshaw, Mark (2011) "Analysts' forecasts: What do we know after decades of work?" Working paper, Boston College.
- Bradshaw, M., Lee, L.F., and Peterson, K. (2014), "The interactive role of difficulty and incentives in explaining the annual earnings forecast walkdown". Working paper
- Bradshaw, Mark T; Richardson, Scott; and Sloan, Richard G. (2001) "Do Analysts and Auditors Use Information in Accruals?" *Journal of Accounting Research*, Vol. 39 No. 1, p.45-74

- Brown, L. (1993) "Earnings forecasting research: its implications for capital markets research". *International Journal of Forecasting* 9: 295–320.
- Burgstahler, David C. & Dichev, Ilia D (1997), "Earnings, adaptation, and equity value", *The Accounting Review*, vol. 72, no. 2, pp. 187-215.
- Clement, M., J. Hales, and Y. Xue (2011) "Understanding analysts' use of stock returns and other analysts' revisions when forecasting earnings" *Journal of Accounting and Economics* 51: 279–299.
- Collins, D. W., Gong, G., and S.P. Hribar. (2003) "Investor sophistication and the mispricing of accruals" *Review of Accounting Studies*, 8, 251–276.
- Collins, D.W, & S.P. Hribar (2000) "Earnings-based and accrual-based anomalies: One effect or two?" *Journal of Accounting and Economics* 29: 101–123
- Collins, Daniel W. & Kothari, S.P. (1989), "An analysis of intertemporal and cross-sectional determinants of earnings response coefficients ", *Journal of Accounting and Economics*, vol. 11, no. 2-3, pp. 143-181.
- Cowen, A., B. Groysberg, and P. Healy (2006), "Which types of analyst firms are more optimistic?" *Journal of Accounting and Economics*, vol. 41, pp. 119–146.
- Dargenidou, C., McLeay, S., and Raonic, I. (2006) "Expected Earnings Growth And The Cost Of Capital: An Analysis Of Accounting Regime Change In The European Financial Market", Abacus, Journal of Accounting, Finance and Business Studies, v.3-4, Pages 388 – 414
- Das, S., Levine C.B., and Sivaramakrishnan K. (1998), "Earnings predictability and bias in analysts' earnings forecasts", *The Accounting Review*, vol. 73, pp. 277-294.
- DeBondt, Werner M., & Thaler, Richard F. (1987), "Further evidence of investor overreaction and stock market seasonality", *Journal of Finance*, 42, 557-582.
- DeBondt, Werner M., & Thaler, Richard F. (1990), "Do security analysts overreact?" *American Economic Rev.*, vol. 80, pp. 52–57.
- Dechow, P. (1994), "Accounting earnings and cash flows as measures of firm performance: The role of accounting accruals," *Journal of Accounting and Economics*, 18, pp. 3–42.
- Dechow, P., & Richard Sloan. (1997) "Returns to contrarian investment strategies: Tests of naïve expectations hypotheses". *Journal of Financial Economics* 43 (1): 3-27.
- Dechow, P.M., Kothari, S.P., Watts, R.L., (1998) "The relation between earnings and cash flows". *Journal of Accounting and Economics* 25,133–168.

- Doyle, J.T., Lundholm, R.J. and Soliman, M.T., (2003), "The predictive value of expenses excluded from pro forma earnings". *Review of Accounting Studies*, vol. 8, pp. 145-174.
- Drake, Mihael & Myers, Linda (2011) "Analysts' accrual-related over-optimism: do analyst characteristics play a role?" *Review of Accounting Studies*, 16(4), p.59-88.
- Easterwood, John C. & Stacey R. Nutt (1999), "Inefficiency in analysts' earnings forecasts: systematic misreaction or systematic optimism?" *Journal of Finance*, vol. 54(5), pp. 1777–1797.
- Easton, P. & Monahan, S. (2005), "An evaluation of the reliability of accounting based measures of expected returns: A measurement error perspective," *The Accounting Review*, vol. 80, pp. 501–538.
- Elgers, P., Lo, M., Pfeiffer, R., (2001) "Delayed security price adjustments to financial analysts\_forecasts of annual earnings". *The Accounting Review* 76 (4), 613–632.
- Elgers, P., M. Lo, and R. Pfeiffer, Jr. (2003), "Analysts' vs. investors' weightings of accruals in forecasting annual earnings," *J. Accounting Public Policy*, vol. 22, pp. 255–280.
- Feltham, Gerald A. & Ohlson, James A. (1995), "Valuation and Clean Surplus Accounting for Operating and Financial Activities", *Contemporary Accounting Research*, vol. 11, no. 2, pp. 689-731.
- Financial Accounting Standards Board (FASB) (1984), Statement of Financial Accounting Concepts No. 5(SFAC 5), Recognition and Measurement in Financial Statements of Business Enterprises, FASB, Stamford.
- Fama, Eugene F. & James D. Macbeth (1973) "Risk, Return, and Equilibrium: Empirical Tests." The Journal of Political Economy, vol. 81, No. 3, pp.607-636.
- Francis, J., & D. Philbrick (1993) "Analysts' decisions as products of a multi-task environment". *Journal of Accounting Research* 31 (2): 216-30.
- Frankel, R., & Lee, C., (1998). "Accounting valuation, market expectations, and cross-sectional stock returns". *Journal of Accounting and Economics* 25 (3), 283–319.
- Frankel, R., S. Kothari, and J. Weber (2006) "Determinants of the informativeness of analyst research". *Journal of Accounting and Economics* 41: 29–54.
- Fried, D. & D. Givoly (1982), "Financial analysts' forecasts of earnings: a better surrogate for market expectations," *Journal of Accounting and Economics*, vol. 4, pp. 85–107.
- Givoly, D., & J. Lakonishok (1984) "Properties of analysts' forecasts of earnings: A review and analysis of the research" *Journal of Accounting Literature* 3: 117-152.

- Griffin, P., (1976) "Competitive information in the stock market: an empirical study of earnings, dividends and analysts forecasts". *Journal of Finance* 31 (2), 631–650.
- Groysberg, B., P. Healy, and D. Maber (2011) "What drives sell-side analyst compensation at high-status investment banks?" *Journal of Accounting Research* 49, 969-1000.
- Gu, Z., and Wu, J. (2003), "Earnings skewness and analyst forecast bias". *Journal of Accounting and Economis*, vol. 35, pp. 5-29.
- Hanlon, M. (2005) "The persistence and pricing of earnings, accruals, and cash flows when firms have large book-tax differences". *The Accounting Review*, 80, 137–166.
- Hayn, Carla (1995), "The information content of losses", Journal of Accounting & Economics, vol. 20, no. 2, pp. 125-153.
- Healy, P (1985), "The effect of bonus schemes on accounting decisions," *Journal of Accounting and Economics*, vol. 7, pp. 85–107.
- Healy, P.M. & Wahlen, J.M. (1999)"A review of the earnings management literature and its implications for standard setting" Accounting Horizons 13-4, pp. 365–384
- Helbok, G. & M. Walker (2004) "On the Nature and Rationality of Analysts' Forecasts Under Earnings Conservatism", *British Accounting Review*, 36-1, pp.45-77
- Hollie, D., Shane, P., and Zhao, Q (2013), "The Role of Financial Analysts in Stock Market Efficiency with Respect to Annual Earnings and its Cash and Accrual Components," *Working Paper, Louisiana State University*
- Hong, H., & J. Kubik (2003) "Analyzing the analysts: Career concerns and biased earnings forecasts". *The Journal of Finance* 58 (1): 313-351
- Hong, H., J. Kubik, and A. Solomon (2000), "Security analysts' career concerns and herding of earnings forecast". *Journal of Economics*, vol. 31, no. 1, pp. 121–144.
- Ke, B. And Yu, Y. (2006), "The effect of issuing biased earnings forecasts on analysts" access to management and survival". *Journal of Accounting Research*, vol 44(5), pp. 965-999.
- Keskek, Sami & Tse, Enyo (2013), "Did Analysts Contribute to the Disappearance of the Accrual Anomaly?" University of Arkansas working paper
- Khan, Muzaffar & Watts, Ross L. (2009) "Estimation and empirical properties of a firm-year measure of accounting conservatism", *Journal of Accounting and Economics* 48.2-3 (2009): 132-150.
- Kim, Jae, Alexndre Nekrasof, Pervin Schrof, and Andreas Simon (2013), "Do analysts understand the valuation implications of accounting conservatism when forecasting target prices?", *http://www.business.uts.edu.au/accounting/pdfs/conf13kim.pdf*,

- Kothari, S.P. (2001), "Capital markets research in accounting", *Journal of Accounting and Economics*, vol. 3, no. 1-3, pp. 105-231.
- Kothari, S., E. Loutskina, and Valeri. Nikolaev (2007) "Agency theory of overvalued equity as an explanation for the accrual anomaly". MIT, CentER Discussion Paper Series No.103
- Konstantinidi, Sonia; Kraft, Arthur G; Pope, Peter F.(2012) "Asymmetric persistence and the pricing of accruals and cash flows", working paper, Cass Business School, available from http://recanati-bs.tau.ac.il/Eng/\_Uploads/dbsAttachedFiles/Kraft.pdf
- Kraft, Arthur. G., A. J. Leone, and C. E. Wasley (2006) "An Analysis of the Theories and Explanations Offered for the Mispricing of Accruals and Accrual Components." *Journal of Accounting Research* vol.44, pp.297–339.
- Kraft, Arthur G., A. J. Leone, and C. E. Wasley (2007) "Regression-Based tests of the market pricing of accounting numbers: The Mishkin test and ordinary least squares". *Journal* of Accounting Research 45: 1081–1114.
- Lev, Baruch & Sougiannis, Theodore (1996), "The Capitalization, Amortization and Value-Relevance of R&D," *Journal of Accounting and Economics*, vol. 21, pp. 107–138.
- Lin, H., & M. McNichols. (1998) "Underwriting relationships, analysts' earnings forecasts and investment recommendations" *Journal of Accounting and Economics* 25: 101-127.
- Lintner, J. (1965) "The valuation of risky assets and the selection of risky investment in stock portfolios and capital budgets" *Review of Economics and Statistics* vol.47 (1), pp13-37.
- Ljungqvist, A., F. C. Marston, L. T. Starks, K. D. Wei, and H. Yan (2007) "Conflicts of Interest in Sell-Side Research and the Moderating Role of Institutional Investors". *Journal of Financial Economics* 85: 420–56.
- Loh, R. & G. Mian (2006). "Do accurate earnings forecasts facilitate superior investment recommendations?" *Journal of Financial Economics*, 80: 455-483.
- Mashruwala, C., S. Rajgopal, and T. Shevlin (2006) "Why is the accrual anomaly not arbitraged away? The role of idiosyncratic risk and transactions costs". *Journal of Accounting and Economics* 13: 341–363.
- McNichols, M., & R O'Brien (1997) "Self-selection and analyst coverage". *Journal of Accounting Research* 35 (3): 167-99.
- McNichols, Rajan, & Reichelstein (2010), "Decomposition of the Market-to-Book Ratio: Theory and Evidence," *Stanford University, Working Paper Series, Available From*

<u>*Http://web-docs.stern.nyu.edu/old\_web/accounting/docs/summercamp/mcnichols.</u> rajan. reichelstein. decomp.of.market.to.debt.ratio.pdf.*</u>

- Mendenhall. R.R. (1991), "Evidence on the possible underweighting of earnings-related information," *J. Accounting Research*, vol. 29, pp. 170–179.
- Michaely, R., & K. Womack. (1999). "Conflict of interest and the credibility of underwriter analyst recommendations". *The Review of Financial Studies* 12 (4): 653-686.
- Mikhail, M., B.Walther, and R. Willis. (1997) "Do security analysts improve their performance with experience?" *Journal of Accounting Research* 35: 131-157.
- Mikhail, M.B., B.R. Walther & R.H Willis (1999). Does forecast accuracy matter to security analysts? *The Accounting Review*, 74, 185-200.
- Miller, M. & Modigliani, F. (1958), "The Cost of Capital, Corporation Finance and the Theory of Investment," *American Economic Rev.*, vol. 48, pp. 261–297.
- Miller, M. & Modigliani, F. (1961), "Corporate Income Taxes and the Cost of Capital: A Correction," *American Economic Rev.*, vol. 53, pp. 433–443.
- Mishkin, F. (1983) "A Rational Expectations Approach to Macroeconometrics: Testing Policy Effectiveness and Efficient Markets Models". Chicago, IL: University of Chicago Press for the National Bureau of Economic Research.
- Mohanram, Partha (2013), "The Role of Analysts' Cash Flow Forecasts in the Decline of the Accruals Anomaly," *Contemporary Accounting Research (forthcoming)*.
- Nissim, Doron & Penman, Stephen (2003), "Financial Statement Analysis of Leverage and How It Informs About Profitability and Price-to-Book Ratios," Review of Accounting Studies, 8, 531–560,
- O'Brien, P. (1988) "Analysts' forecasts as earnings expectations". *Journal of Accounting and Economics* 10(1): 159-93.
- O'Hanlon, J. (1993) "Commentary on: Lawrence D. Brown 'Earnings forecasting research, its implications for capital markets research" *International J.of Forecasting* 9: 321-323.
- Pae, Jinhan & Thornton, Daniel (2010) "The Association Between Accounting Conservatism and Analysts' Forecast Inefficiency", Asia-Pacific Journal of Financial Studies, vol.39, 2, pp.171-197
- Palepu,K.G, Healy,P.M., Bernard, V.L. (2000), "Business Analysis and Valuation," 2nd Ed., Southwestern College Publishing.
- Penman, S. (2007), Financial Statement Analysis and Security Valuation. Third edition. New York, NY: McGraw-Hill,

- Penman, S., & X. Zhang (2002) "Accounting conservatism, the quality of earnings, and stock returns" *The Accounting Review*, vol. 77, no.2, pp. 237-264.
- Penman, S., & X. Zhang (2013), "Book Rate of Return, Risk, and the Information Conveyed by Conservative Accounting," *Working Paper Series, Available From* <u>Http://recanatibs.tau.ac.il/eng/\_uploads/dbsattachedfiles/zhang.pdf</u>
- Penman, S., Richardson, S., and Irem Tuna (2007) "The Book-to-Price Effect in Stock Returns: Accounting for Leverage" *Journal of Accounting Research*, 45, 2, pp.427– 467
- Petersen, Mitchell (2009), "Estimating Standard Errors in Finance Panel Data Sets: Comparing Approaches," Oxford Journals Economics & Social Sciences Rev. Financial Studies, vol. 22, no. 1, pp. 435–380.
- Raedy, J., P. Shane, and Y. Yang. (2006), "Horizon-dependent underreaction in financial analysts' earnings forecasts," *Contemporary Accounting Research*, vol.23, pp.291– 322.
- Ramnath, S., S. Rock, and P. Shane (2008) "A review of research related to financial analysts' forecasts and recommendations". *Foundations and Trends in Finance* 24: 311–421.
- Richardson, S. A, Teoh, S. H., Wysocki, P. D. (2004). "The walk-down to beatable analyst forecasts: The role of equity issuance and insider trading incentives". *Contemporary Accounting Research*, 21, 885–924.
- Richardson, Scott; Irem Tuna and Peter Wysocki (2010), "Accounting anomalies and fundamental analysis: A review of recent research advances," *Journal of Accounting and Economics*, vol. 50, no. 2-3, pp. 410–454
- Richardson, Scott; Soliman, Mark; Sloan, Richard G; and Tuna, Irem (2005), "Accrual reliability, earnings persistence and stock prices," *Journal of Accounting and Economics*, vol. 39, pp. 437–485.
- Roychowdhury, Sugata & Wats, Ross L. (2007), "Asymmetric timeliness of earnings, market-to-book and conservatism in financial reporting," *Journal of Accounting and Economics*, vol. 44, no. 1-2, pp. 2–31.
- Schipper, K. (1991) "Analysts' forecasts". Accounting Horizons 5: 105–121.
- Schwert, G. (2003) "Anomalies and market efficiency", Chapter 15. In George Constantinides, Milton Harris, and Rene' Stulz (Eds.) Handbook of the economics of finance (pp. 937–972). North-Holland.

- Schwert, W. (2001) "Anomalies and market efficiency", in Handbook of the Economics of Finance, chapter 17.
- Sharpe, W. (1964) "Capital asset prices: A theory of market equilibrium under conditions of risk" *The Journal of Finance* vol.19, no.3, pp.425-442.
- Simon, A., & Curtis, A. (2011), "The use of earnings forecasts in stock recommendations: Are accurate analysts more consistent?" J. Business Finance & Accounting, vol. 38, no. 12, pp. 119–144.
- Sloan, Richard G. (1996), "Do Stock Prices Fully Reflect Information in Accruals and Cash Flows About Future Earnings?" *The Accounting Rev.*, vol. 71, no. 3, pp. 289–315.
- Stober, T.L. (1992), "Summary financial statement measures and analysts' forecasts of earnings," *Journal of Accounting and Economics*, vol. 15, pp. 347–372.
- Subramahyam, K.R. & Wild, John J. (1996), "Going Concern Status, Earnings Persistence, and Informativeness of Earnings", *Contemporary Accounting Research*, vol. 13, no. 1, pp. 251-273.
- Zach, T., (2007), "Evaluating the "accrual-fixation" hypothesis as an explanation for the accrual anomaly." Washington University, AAA 2005 FARS Meeting Paper
- Zhou, J., (2012), "Do analysts act strategically to induce management guidance? Theory and evidence". National University of Singapore working paper.

# Appendix

Dependent Variables						
ROA	Operating income after depreciation (Compustat Item OIADP, #178) deflated by average assets (Compustat Item AT, #6).					
Ferror	Realized earnings for year t+1 less forecasted consensus (median) earnings in month <i>s</i> ( <i>s</i> =1, 2, 3,, 12), scaled by the stock price in month <i>s</i> =1. Month <i>s</i> =1 is the first month following prior year's earnings announcement. <i>Ferror s</i> , <i>t</i> +1 = [Actual EPS <i>t</i> +1 – <i>Forecast EPS s</i> , <i>t</i> +1] / $P_{1,t}$					
Independent Variables						
TACC	Change in non-cash assets - change in liabilities, deflated by average assets.					
ΔΟΡΑϹ	Operating accruals is the change in non-cash working capital $(\Delta WC_t)$ plus change in net non-current operating assets ( $\Delta NCO_t$ ), deflated by average assets.					
ΔWC	Working capital accruals is the change in net working capital $= WC_t - WC_{t-1}$ . WC is current operating assets (COA) less operating liabilities (COL). COA=current assets (Compustat Item ACT, #4) - cash and short term investments (Compustat Item CHE, #1), and COL=current liabilities (Compustat Item LCT, #5) - short term debt (Compustat Item DLC, #34).					
ΔΝCΟ	Non-current operating accruals is the change in net non- current operating assets = $NCO_t - NCO_{t-1}$ . NCO is = non- current operating assets (NCOA) - non-current op.liabilities (NCOL). NCOA=total assets (Compustat Item AT, #6) - current assets (Compustat Item ACT, #4) - investments and advances (Compustat Item IVAO, #32), and NCOL=total liability (Compustat Item LT, #181) - current liabilities (Compustat Item LCT, #5) - short term debt (Compustat Item DLC, #34) - long term debt (Compustat Item DLTT, #9)					
ΔFIN	Financing accruals is the change in net financial assets = $FIN_t$ - $FIN_{t-1}$ . $FIN$ =financial assets ( $FINA$ ) - financial liabilities ( $FINL$ ). $FINA$ =short term investments (STI) (Compustat Item IVST, #193) + long term investments (LTI) (Compustat Item IVAO, #32). $FINL$ = long term debt (Compustat Item DLTT, #9) + short term debt (Compustat Item DLC, #34) + preferred stock (Compustat Item UPSTKC, #130)					

Conditional accounting conservatism proxy varying across

firms and years developed by Khan & Watts (2009) and derived from Basu's (1997) asymmetric timelines of earnings measure:

$$C\_Score_{it} \equiv \beta_4 = \gamma_1 + \gamma_2 Size_{it} + \gamma_3 \frac{M_{it}}{B_{it}} + \gamma_4 lev_{it}$$

The parameters are calculated using the following annual cross sectional regression model

$$X_t = \beta_1 + \beta_2 D_i + \beta_3 R_i (\mu_1 + \mu_2 Size_i + \mu_3 M/B_i + \mu_4 lev_i D_i)$$

$$+\beta_4 D_i R_i (\gamma_1 + \gamma_2 Size_i + \gamma_3 M/B_i + \gamma_4 lev_i) + \delta_1 Size_i$$

 $+\delta_2 M/B_i + \delta_3 Lev_i + \delta_4 D_i Size_i + \delta_5 D_i M/B_i + \delta_6 D_i Lev_i + e_{it}$ 

Where X is earnings before extraordinary items (Compustat Item IB, #18) at time t deflated by market value at time t-1, where market value is calculated as the share price (Compustat item PRCC\_F, #199) multiplied by the common shares outstanding (Compustat item CSHO, #25). R represents annual buy and hold returns inclusive of dividends and other distributions at time t, where accumulation period starts in the fourth month after the fiscal year end t-1, and continues for the next twelve months up to third month of fiscal year t+1. D is set to 1 if R < 0 and zero otherwise. The coefficient  $\beta_4$  measures the incremental timeliness for bad news over good news, or conservatism. E/P is income at time *t* deflated by market value at time *t*-1. Size is the natural log of market value at time t, leverage is measured as long term debt (Compustat Item DLTT, #9) plus short term debt (Compustat Item DLC, #34)] divided by the market value at time t. M/B is calculated as market value at time t divided by the book value of equity at time t. Following Khan & Watts (2009), all firm years with missing data, negative total assets and book values are eliminated in estimations. Firms with price per share less than \$1 are eliminated, and all variables are winsorised to 1% and 99%.

*Hidden\_Reserves* Unconditional accounting conservatism proxy by Penman & Zhang, (2002) deflated by average assets. It is calculated as:

 $Hidden_Reserves_t = R\&Dres_t + ADVres_t + LIFOres_t$ 

R&Dres is research and development reserve which is unamortised balance of R&D expenditures (Compustat Item XRD, #46) that would have appeared on balance sheet if it had been capitalised and amortised at a straight line rate of 20%, assuming a uniform distribution.

$$\begin{split} R\&Dres_{it} &= 0.9R\&D_{it} + 0.7R\&D_{it-1} + 0.5R\&D_{it-2} \\ &+ 0.3R\&D_{it-3} + 0.1R\&D_{it-4} \end{split}$$

ADVres is advertisement reserve calculated using advertisement expenditures (Compustat Item XAD, #45) assuming a useful life of two years, and providing more benefits when first initiated

$$ADres_{it} = ADV_{it} + 1/3ADV_{it-1}$$

*LIFOres* is LIFO reserves reported in the inventory footnotes in financial reports (Compustat Item LIFR, #240).

<b>Control Variables</b>	
Size	Natural log of market value of equity. Market value is calculated as the share price (Compustat item PRCC_F, #199) multiplied by common shares outstanding (Compustat item CSHO, #25)
B/P	Book value of equity divided by market value of equity. Book value of equity = Common ordinary equity total (Compustat Item CEQ, #60) + Preferred treasury stock Current Assets (Compustat Item TSTKP, #227) + Preferred dividends in arrears (Compustat Item DVPA, #242).
Market Beta	Calculated through the 60 month rolling regressions using the market model $(Ret_{it} - R_f) = \alpha + \beta_i (Ret_{mt} - R_f) + \epsilon_{it}$ <i>Ret</i> is the CRSP monthly buy and hold returns for 12 month for stock <i>i</i> at time <i>t</i> , $R_f$ is risk the free rate, $(Ret_{mt} - R_f)$ is the equity risk premium of the market portfolio. $R_f$ is obtained from the US Federal Reserve, H15 report as the 10-year US Treasury bond rate for the relevant year. $Ret_{mt}$ is the CRSP monthly value weighted return on a market portfolio cumulated over 12 months.
Past_return	Size adjusted past returns, calculated as the sum of 12-month buy and hold CRSP, NYSE, AMEX/NASDAQ stock returns from nine months before fiscal year-end $t$ to three month after fiscal year-end $t$ minus the corresponding value-weighted average returns for all firms in the same size-matched decile (returns are inclusive of dividends and other distributions). To form deciles market values are ranked annually, and assigned in equal numbers to ten portfolios. In tests, the variable <i>is</i> winsorised to %1 and %99.

E/P	Earnings to price ratio calculated as operating income after depreciation (Compustat Item OIADP, #178) at time $t$ deflated by market value of equity at time $t$ -1.
Loss_D	Loss dummy set to 1 if the past year earnings is negative and zero otherwise.
Leverage	Long term debt (Compustat Item DLTT, #9) plus short term debt (Compustat Item DLC, #34)] divided by the market value at time <i>t</i> .

Table 1 Descriptive statistics and correlations for ROA, accrual components, and conservatism measures

	mean	std.dev.	25%	median	75%
$ROA_{t+1}$	0.045	0.214	0.007	0.08	0.14
$ROA_t$	0.043	0.186	0.002	0.076	0.136
$TACC_t$	0.051	0.195	-0.021	0.037	0.109
$4OPAC_t$	0.063	0.195	-0.027	0.041	0.135
$\Delta FIN_t$	-0.012	0.176	-0.071	-0.002	0.048
$\Delta WC_t$	0.015	0.106	-0.024	0.008	0.052
$\Delta NCO_t$	0.048	0.159	-0.015	0.021	0.084
C_Score t	0.013	0.115	-0.052	0.012	0.081
Hidden_Reserves,	0.163	0.190	0.035	0.098	0.218

#### PANEL B: Correlation matrix—Pearson (above diagonal) and Spearman (below diagonal)

	$ROA_{t+1}$	$ROA_t$	$TACC_t$	$\Delta OPAC_t$	$\Delta FIN_t$	$\Delta WC_t$	$\Delta NCO_t$	$C_S_t^d$	$H_R_t$
$ROA_{t+1}$	-	0.75 ***	0.13 ***	0.09 ***	0.05 ***	0.01 ***	0.04 ***	-0.24 ***	-0.16 ***
$ROA_t$	0.79 ***	-	0.22 ***	0.18 ***	0.05 ***	0.20 ***	0.08 ***	-0.27 ***	-0.17 ***
$TACC_t$	0.23 ***	0.38 ***	-	0.69 ***	0.45 ***	0.40 ***	0.47 ***	-0.10 ***	-0.02 ***
$\Delta OPAC_t$	0.13 ***	0.27 ***	0.60 ***	-	-0.45 ***	0.60 ***	0.84 ***	-0.08 ***	-0.06 ***
$\Delta FIN_t$	0.09 ***	0.08 ***	0.29 ***	-0.47 ***	-	-0.22 ***	-0.41 ***	-0.02 ***	0.04 ***
$\Delta WC_t$	0.12 ***	0.23 ***	0.41 ***	0.63 ***	-0.27 ***	-	0.07 ***	-0.01 ***	-0.02 ***
$\Delta NCO_t$	0.11 ***	0.22 ***	0.47 ***	0.80 ***	0.41 ***	0.16 ***	-	-0.12	-0.07 ***
$C\_Score_t$	-0.02 ***	-0.02 ***	-0.08 ***	-0.06 ***	-0.01 ***	-0.01 ***	-0.06	-	0.05
Hidden_Reserves t	-0.39 ***	-0.43 ***	-0.05 ***	-0.08 ***	0.03 ***	-0.05 ***	-0.07 ***	0.00	-

Earmings and accruals sample consists of 142,633 firm-year observations, while Hidden\_Reserves and C\_Score samples consist of 98,196 and 96,324 firm-year observations respectively from 1976 to 2013. All earnings and accrual variables are deflated by average total assets and winsorised to +1 and -1, while C\_Score and Hidden\_Reserves are winsorised at 15 and 99%. Table A in the Appendix reports detailed variable definitions. \*\*\* denotes statistical significance at the 1% level.

 Table 2

 Descriptive statistics and correlations for ROA and the extended accrual decomposition

	mean	std.dev.	25%	median	75%
$\Delta COA_t$	0.004	0.132	-0.01	0.022	0.081
$\Delta COL_t$	0.025	0.09	-0.009	0.015	0.051
$\Delta NCOA_t$	0.055	0.163	-0.012	0.025	0.091
$\Delta NCOL_t$	0.006	0.049	-0.001	0.001	0.011
$\Delta STI_t$	0.007	0.105	0	0	0
$\Delta LTI_t$	0.002	0.047	0	0	0
$\Delta FINL_t$	0.021	0.141	-0.023	0	0.051

	ROA <sub>t+1</sub>	ROA <sub>t</sub>	$\Delta \text{COA}_{t}$	$\Delta \text{COL}_{t}$	$\Delta NCOA_t$	$\Delta \text{NCOL}_{t}$	$\Delta STI_t$	$\Delta LTI_t$	$\Delta \text{FINL}_{\text{t}}$
$ROA_{t+1}$	-	0.75 ***	0.11 ***	0.04 ***	0.05 ***	0.02 ***	0.03 ***	0.02 ***	-0.03 ***
$ROA_t$	0.79 ***	-	0.16 ***	0.00	0.09 ***	0.01 ***	0.04 ***	0.02 ***	-0.02 ***
$\Delta COA_t$	0.20 ***	0.31 ***	-	0.60 ***	0.29 ***	0.08 ***	0.01	0.01 ***	0.33 ***
$\Delta COL_t$	0.15 ***	0.18 ***	0.57 ***	-	0.31 ***	0.07 ***	0.09 ***	0.03 ***	0.20 ***
$\Delta NCOA_t$	0.14 ***	0.25 ***	0.38 ***	0.35 ***	-	0.23 ***	-0.01	-0.01 ***	0.51 ***
$\Delta NCOL_t$	0.15 ***	0.19 ***	0.14 ***	0.11 ***	0.31 ***	-	0.01 ***	0.06 ***	0.03 ***
$\Delta STI_t$	0.07 ***	0.09 ***	-0.02 ***	0.08 ***	-0.02 ***	0.02 ***	-	-0.02 ***	0.03 ***
$\Delta LTI_t$	0.02 ***	0.04 ***	0.03 ***	0.04 ***	0.02 ***	0.06 ***	0.00	-	0.08 ***
$\Delta FINL_t$	-0.04 ***	-0.04	0.32 ***	0.17 ***	0.05 ***	0.11 ***	-0.02 ***	0.05 ***	-

The sample consists of 142,821 firm-year observations from 1976 to 2013. All earnings and accrual variables are deflated by average assets and winsorised to +1 and -1. Variable definitions are provided in Table A in the Appendix. \*\*\* denotes statistical significance at the 1% level.

	Table	e 3	
Accrual	components	persistence tests	

PANEL A: Cash flows and total accruals persistence								
$ROA_{it+1} = \rho_0 + \rho$	$\rho_1 ROA_{it} + \rho_2 TACC$	$\theta_{it} + \theta_{it+1}$						
	intercept	ROA	TACC	$R^2$				
mean coef.	0.008	0.797 ***	-0.068 ***	0.632				
		99.11	-16.15					

#### PANEL B: Operatig and financial accruals persistence

 $\textit{ROA}_{it+1} = \rho_0 + \rho_1 \textit{ROA}_{it} + \rho_2 \Delta \textit{OPAC}_{it} + \rho_3 \Delta \textit{FIN}_{it} + \vartheta_{it+1}$ 

	intercept	ROA	$\Delta OPAC$	$\Delta FIN$	$R^2$
	-		Low	High	
mean coef.	0.007	0.795 ***	-0.070 ***		0.631
			-16.35		
mean coef.	0.005	0.781 ***		0.005	0.629
				0.26	
mean coef.	0.009	0.801 ***	-0.089 ***	-0.047 ***	0.632
			-17.30	-13.49	

## PANEL C: Working capital, non-current operating and financial accruals persistence

 $\textit{ROA}_{it+1} = \rho_0 + \rho_1 \textit{ROA}_{it} + \rho_2 \Delta \textit{WC}_{it} + \rho_3 \Delta \textit{NCO}_{it} + \rho_4 \Delta \textit{FIN}_{it} + \vartheta_{it+1}$ 

	intercept	ROA	∆WC	ΔΝCΟ	⊿FIN	$R^2$
mean coef.	0.007	0.791 ***	-0.122 *** -16.09			0.631
mean coef.	0.008	0.782 ***		-0.051 *** -10.79		0.625
mean coef.	0.005	0.777 ***		10.77	0.002 0.26	0.629
mean coef.	0.009	0.804 ***	-0.137 *** -19.59	-0.065 *** -12.32	-0.045 *** -11.78	0.634
Persistence Order			Lowest	Medium	Highest	

#### PANEL D: Individual accrual components persistence

	intercept	ROA	(-)∆COL	ΔCOA	(-)ANCOL	ΔΝϹΟΑ	ΔLTI	(-)∆FINL	⊿STI	$R^2$
mean coef.	0.005	0.776 ***	-0.035 *** -4.03							0.62
mean coef.	0.008	0.786 ***	-4.05	-0.065 *** -11.3						0.63
mean coef.	0.005	0.777 ***		11.5	-0.044 *** -4.52					0.62
mean coef.	0.007	0.782 ***				-0.047 *** -10.17				0.62
mean coef.	0.006	0.78 ***				1011/	-0.037 *** -4.45			0.63
mean coef.	0.005	0.776 ***					1.10	0.010 *** 2.00		0.62
mean coef.	0.006	0.775 ***						2.00	-0.026 *** -5.53	0.62
mean coef.	0.008						-0.059 *** -6.65	-0.052 *** -9.46	-0.034 ***	0.63
Persistence	Order		Lowest	2	3	4	5	/	Highest	

$$\begin{split} ROA_{it+1} &= \rho_0 + \rho_1 ROA_{it} + \rho_2 \Delta COL_{it} + \rho_3 \Delta COA_{it} + \rho_4 \Delta NCOL_{it} + \rho_5 \Delta NCOA_{it} \\ &+ \rho_6 \Delta LTI_{it} + \rho_7 \Delta FINL_{it} + \rho_8 \Delta STI_{it} + \vartheta_{it+1} \end{split}$$

Standard errors are clustered by firm and year using the Petersen (2009) approach. The sample consists of 142,821 firm-year observations fot 1976-2013, all earnings and accrual variables are deflated by average total assets and winsorised to +1 and -1. Variable definitions are provided in Table A of the Appendix. \*\*\* denotes statistical significance at the 1% level.

Table 4								
Forecast errors	and accrual	components						

		m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
Intercept		-0.015	***-0.014 *	*** -0.12	***-0.011 **	* -0.01 *	**-0.008 **	**-0.007 *	**-0.006 **	-0.005 *	**-0.004 *	**-0.003 **	*-0.003 **
	Persis	tence Or	der <sup>(a)</sup>										
4WC	1 Low	-0.039	***-0.038 *	***-0.033	***-0.027 **	*-0.025 *	**-0.026 **	**-0.023 *	**-0.019 **	-0.015 **	**-0.013 *	**-0.011 **	*-0.010 *
		-8.63	-8.85	-5.98	-6.19	-6.27	-9.12	-8.52	-7.11	-6.73	-5.25	-3.82	-4.46
4NCO	2	-0.001	0	-0.01	-0.007	-0.01	0	-0.01	-0.004	0	0	0	0
		-0.33	-1.00	-1.22	-1.06	-1.13	-0.99	-0.97	-0.78	-0.47	-0.86	-0.81	-0.74
1FIN	3 High	0.017	*** 0.015 *	*** 0.012	*** 0.011 **	* 0.01 *	** 0.01 **	** 0.009 *	** 0.008 ***	• 0.007 *	** 0.005 *	** 0.005 **	* 0.005 *
		5.46	5.03	4.38	3.82	3.76	4.15	3.84	3.96	4.27	4.00	3.56	3.88
PANEL B:	Forecast	errors a	and total ac	cruals									
$error_{s,it+1}$	$= \beta_0 + \beta$	1TACC <sub>it</sub>	$+\varepsilon_{it+1}$										
Month		m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
ntercept		-0.017	***-0.015 *	***-0.013	***-0.012 **	* -0.01 *	**-0.009 **	**-0.008 *	**-0.007 ***	-0.005 **	**-0.004 *	**-0.003 **	*-0.003 **
TACC		0.002	0	0	0	0	0	0	0	0.001	0	0	0

The number of firm-year observations ranges from 45,145 to 48,142 across twelve months from 1976 to 2013 for which consensus analyt earning forecasts and actual earnings are available on the IBES summary statistics file. Standard errors clustered by firm and year using the Petersen (2009) approach. Variable definitions are provided in Table A of the Appendix. \*\*\* denotes statistical significance at the 1% level.

<sup>(a)</sup> indicates the persistence order of accrual components obtained from the corresponding multivariate earnings persistence regression from Table 3.

Ferror <sub>s,it</sub>	$\mu_1 = \beta_0$	$+\beta_1 \Delta COL$	$_{it} + \beta_2 \Delta C C$	$OA_{it} + \beta_3 \Delta$	$NCOL_{it} +$	$\beta_4 \Delta NCOA$	$\varepsilon_{it} + \varepsilon_{it+1}$						
		m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
Intercept		-0.016 *	**-0.014 *	**-0.012 *	**-0.011 *	**-0.010 *	**-0.008 *	**-0.008 *	***-0.007 *	**-0.005 *	** -0.004 *	**-0.003 *	**-0.003 ***
	Persi	stence Ord	er <sup>(a)</sup>										
(-)∆COL	1 Lov	v -0.062 *											**-0.018 ***
		-10.42	-9.63	-5.09	-5.27	-6.25	-6.54	-6.34	-4.89	-4.14	-3.5	-2.83*	-3.46
ΔCOA	2				***-0.025 * -5.47								***-0.009 **** -4.03
(-)ANCOL	3	0.032 *	**-0 038 *	** 0 029 *	**-0.044 *	* 0.049 *	* -0 025 *	** -0.056	-0.053	0.016 *	**-0 045	-0.044	-0.039
(921/002	5				-2.27				-1.57		-1.32	-1.28	-1.19
∆NCOA	4 Hig	<b>h</b> -0.002	-0.003	-0.005	-0.006	-0.006	-0.002	-0.004	-0.003	-0.00	-0.003	-0.002	-0.002
		-0.71	-1.22	-1.32	-1.03	-0.99	-0.69	-0.79	-0.66	-0.11	-0.69	-0.73	-0.62

 Table 5

 Forecasts optimism and accrual persistence

The number of firm-year observations ranges from 45,145 to 48,142 across twelve months from 1976 to 2013 for which consensus analyt earning forecasts and actual earnings are available on the IBES summary statistics file. Standard errors clustered by firm and year using the Petersen (2009) approach. Variable <sup>(a)</sup> indicates the persistence order of accrual components obtained from the corresponding multivariate earnings persistence regression from Table 3.

 Table 6

 Forecast errors, accrual components and controls

PANEL A: m1	(month one	e) OLS regr	essions of	forecast	errors on ind	ividual accru	al compone	nts and con	ntrols	
_	$\sum_{n=1}^{n}$	$\sum_{m=1}^{m}$	$\sum_{i=1}^{n}$			ended accrual		s)		
$Ferror_{1,it+1} = \beta_0 + $	$\sum_{k} \beta_k ACC_{ii}$	$+\sum_{i}\beta_{k+j}X_{it}$	$+\sum_{i}\delta_{i}$ Ind	$t + \varepsilon_{it+1}$		trol variable.			1 1005	
	<i>k</i> =1	<i>j</i> =1	(=1		z = 48 (in	dustries, cate	gorised by F	ama & Fr	ench, 1997)	
		No	W		s (each regres	sion includes o			e)	All
		Controls	beta	B/P	Size	leverage	Loss_D	E/P	Past_Ret	Controls
		nce Order (a								
(-)∆COL	1 Low					** -0.060 ***				
		-10.42	-12.63	-11.10	-11.28	-12.31	-14.90	-9.97	-4.75	-4.55
∆COA	2	-0.037 ***	-0.027 *	** -0.045	*** -0.016 *	** -0.040 ***	-0.046 ***	-0.044 **	** -0.033 ***	* -0.060 ***
		-7.26	-7.86	-14.38	-5.22	-10.93	-15.71	-8.21	-5.95	-11.54
(-)ANCOL	3	-0.032 ***	-0.024 *	** -0.037	*** -0.014 *	** -0.028 ***	• -0.019 ***	-0.017 **	** 0.005	-0.027 ***
		-4.02	-3.39	-4.74	-2.16	-3.78	-3.38*	-2.87	1.65	-3.12
ΔΝϹΟΑ	4	-0.002	-0.001	-0.012	*** -0.005 *	* -0.010 ***	-0.004	0.001	-0.031 ***	* -0.017 ***
		-0.71	-0.67	-6.28	-2.16	-5.70	-2.65	0.41	-3.51	-4.96
∆LTI	5	0.006	0.008	• 0	0.002	-0.003	0.008	0.011 **	• 0.011 ***	-0.007
		1.33	1.68	0.09	0.59	-0.48	1.26	1.92	3.96	-0.080
(-)∆FINL	6	0.025 ***	0.030 *	** 0.019	*** 0.027 *	** 0.010 ***	0.020 ***	0.027 **	•• 0.003	-0.012 ***
		5.95	6.68	9.08	12.83	4.87	9.92	6.16	0.47	-3.55
⊿STI	7 High	0.010 ***	0.011 *	** 0.002	0.007 *	** 0.006 ***	0.013 ***	0.012 **	•• 0.029 •••	0.003
	, in the second s	4.23	5.01	1.46	3.84	2.74	6.24	3.81	5.72	0.084
Intercept			-0.003 *	** -0.029	*** 0.006 *	** -0.016 ***	-0.058 ***	0.162 **	•• 0.023 •••	∗ all
-			-5.09	-44.64	49.25	-34.61	-92.88	15.58	<b>13.78</b> sig	nificant
$R^2$		2.5%	2.8%	6.7%	7.6%	5.3%	17.5%	15.9%	6.8%	34.7%
Observations		48,159	45,729	46,795	48,015	41,124	48,070	47,589	44,853	35,307

The sample period is 1976 and 2013 for which consensus analysts earning forecasts and actual earnings are available on the IBES summary statistics file. Standard errors clustered by firm and year using the Petersen (2009) approach, and industry fixed effects are controlled according to Fama & French (1997) 48 industry classifications. Variable definitions are provide in Table A of the Appendix. \*\*\* and \*\* denote statistical significance at the 1% and 5% levels, respectively.

 $\stackrel{(a)}{=}$  indicates the persistence order of a particular accrual component obtained from the accrual persistence tests reported in Table 3.

 Table 7

 Forecast errors, accrual components and conservatism

Ferror <sub>1,it+1</sub> =			$\sum_{i=1}^{3} \gamma_k D_i * QACC_i$	$t + \varepsilon_{it+1}$				
	ĸ		=1 dden_Reserves				C_Score	
		m1	m2	m3		m1	m2	m3
Intercept		-0.018 ***	-0.016 ***	-0.014 ***		-0.018 ***	-0.016 ***	-0.014 ***
D		0	0	0.001		0	0	0.001
	Persistence	Low		Low conservatism				
QΔWC	1 Low	-0.022 **	-0.022 **	-0.020 **		-0.031 ***	-0.032 ***	-0.026 ***
Q4NCO	2	0.011 **	0.008	0.006		0.004	0.003	-0.003
Q⊿FIN	3 High	0.040 ***	0.035 ***	0.031 ***		0.016 ***	0.014 ***	0.014 ***
		Increme	ental conservatism	n		Increme	ental conservatis	n
D*Q∆WC	1 Low	-0.021 **	-0.019	-0.019		0.007	0.008	-0.003
D*QANCO	2	-0.023 ***	-0.020 ***	-0.021 ***		0	-0.004	0.006
D*Q∆FIN	3 High	-0.029 ***	-0.027 ***	-0.026 ***		0.020 ***	0.013	0.011
Sum of coefficie	ents ( $\delta$ + Y)	High	conservatism			High	conservatism	
<i>Q∆WC+D*Q∆</i> И	WC	-0.043 ***	-0.041 ***	-0.039 ***	- F	-0.024 ***	-0.024 ***	-0.028 ***
Q∆NCO+D*Q∠		-0.012	0.012	0.016		0.003	0	0.004
Q∆FIN+D*Q∆	FIN	0.011 ***	0.009 ***	0.005		0.036 ***	0.028 ***	0.024 ***

The samples consist of 14,207 (14,707) firm-year observations for *Hidden\_reserves* ( $C_Score$ ) between 1976 and 2013. QAWC, QANCO, and QAFIN are quintile rankings for  $\Delta WC$ ,  $\Delta NCO$  and  $\Delta FIN$  based on their respective magnitudes in year t. D is an indicator variable which assignes 0 (1) for the lowest (highest) conservatism quintile. Conservatism quintiles are based on the magnitudes of *Hidden\_reserves* ( $C_Score$ ) for the first three months of year t before the announcement of year t quarterly earnings. Standard errors clustered by firm and year using the Petersen (2009) approach. All variables are defined in Table A in the Appendix. Regressions are run for the first three months before the announcement of quarterly earnings. \*\*\* and \*\* denote statistical significance at the 1% and 5% levels, respectively.

<sup>(a)</sup> indicates the persistence order of accrual componenst obtained from the corresponding multivariate earnings persistence regression from Table 3.

	Table 8	
Forecast errors,	accruals	and cash flows

PANEL A: Operating and financial accruals and cash flow components of earnings

 $\textit{Ferror}_{s,it+1} = \beta_0 + \beta_1 \Delta \textit{OPAC}_{it} + \beta_2 \Delta \textit{FIN}_{it} + \beta_3 \textit{CF}_{it} + \varepsilon_{it+1}$ 

		m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
Intercept		-0.017 ***	-0.014 ***	-0.12 ***	-0.011 ***	-0.009 ***	-0.008 ***	-0.007 ***	-0.006 ***	-0.005 ***	-0.004 ***	-0.003 ***	-0.003 ***
	Persist	ence Order	<i>(a)</i>										
$\Delta OPAC$	1 Low	-0.010 ***	-0.012 ***	-0.012 ***	-0.011 ***	-0.011 ***	-0.008 ***	-0.009 **	-0.007 *	-0.004 **	-0.005	-0.004	-0.004
		-3.66	-4.20	-4.21	-2.82	-2.70	-3.29	-2.18	-1.78	-2.42	-1.53	-1.36	-1.30
∆FIN	2	0.009 ***	0.008 ***	0.006 **	0.006 **	0.005 **	0.005 **	0.005 **	0.004 *	0.004 **	0.003 **	0.003 **	0.002 *
		3.39	2.99	2.43	2.21	2.04	2.20	1.90	1.85	2.20	2.05	2.13	1.91
CF	3 High	0.042 ***	0.035 ***	0.030 ***	0.027 ***	0.025 ***	0.021 ***	0.020 ***	0.016 ***	0.013 ***	0.012 ***	0.011 ***	0.010 ***
		8.58	8.57	7.10	7.70	7.77	6.85	7.48	6.99	6.39	6.63	6.58	6.29

#### PANEL B: Financial accruals and cash flow components

Ferror <sub>s,it+1</sub>	$= \beta_0 +$	$\beta_1 \Delta LT I_{it}$	$+\beta_2 \Delta FINL_i$	$_{t} + \beta_{3}\Delta STI_{i}$	$t + \beta_4 CF_{it} + \varepsilon_{it+1}$
--------------------------	---------------	----------------------------	--------------------------	----------------------------------	--

		m1	m2	m3	m4	m5	m6	m7	m8	5 m9	m10	m11	m12
Intercept		-0.015 *	***-0.014	*** -0.12	***-0.011	*** -0.01	***-0.008	***-0.007	***-0.006	***-0.005	***-0.004	***-0.003 **	*-0.003 ***
	Persist	ence Ord	ler <sup>(a)</sup>										
$\Delta LTI$	1 Low	0.006	0.007	0.006	0.007	0.005	0.093	0.004	0.003	0.004	0.004	0.002	0.002
		1.28	1.50	1.12	1.37	1.04	0.77	0.99	0.68	1.25	1.25	0.62	0.85
∆FINL	2	0.016	*** 0.016	*** 0.014	*** 0.013	*** 0.013	*** 0.011	*** 0.011	*** 0.008	*** 0.013	** 0.012	* 0.012	0.005 ***
		5.02	5.26	4.38	4.35	4.86	5.28	4.47	4.25	1.93	1.67	1.57	2.92
⊿STI	3	0.010 -	*** 0.008	*** 0.007	*** 0.006	** 0.006	0.006	** 0.005	*** 0.006	*** 0.007	*** 0.007	** 0.007 **	0.003 **
		3.72	3.11	2.22	1.68	1.85	1.96	2.07	2.54	2.90	2.19	2.37	2.36
CF	4 High	0.041 ·	*** 0.035	*** 0.030	*** 0.029	*** 0.024	*** 0.020	*** 0.020	*** 0.016	*** 0.012	*** 0.011	*** 0.010 **	* 0.010 ***
	0	8.71	8.65	7.15	7.67	7.79	6.84	7.21	6.82	5.60	6.68	6.54	5.92

The number of firm-year observations ranges from 41,515 to 39,898 across twelve months from 1976 to 2013 for which consensus analyte earning forecasts and actual earnings are available on the IBES summary statistics file. F-tests (untabulated) reveal that the coefficients are statistically different from each other. Standard errors clustered by firm and year using the Petersen (2009) approach (see section 6 for the definition of variables). \*\*\*, \*\*, and \* denote statistical significance at the 1%, 5% and %10 levels, respectively.

<sup>(a)</sup> indicates the persistence order of accrual components obtained from the corresponding multivariate earnings persistence regression from Table 3.

	Table 9           Future stock returns and accruals										
PANEL A: Future stock returns and total accruals											
$Ret_{it+1} = \beta_0 +$	$Ret_{it+1} = \beta_0 + \beta_1 TACC_{it} + \sum_{i=1}^{m} \beta_{j+1} X_{it} + \varepsilon_{it+1} \qquad m=5 \ (control \ variables)$										
	Intercept	TACC	Beta	B/P	E/P	Size <sup>(a)</sup>	Past_Return $R^2$				
mean coef.	-0.074 *** -2.05	-0.173 ** -6.38	** 0.035 ** 2.15	* 0.037 * 1.78	0.67 * 10.52	** -0.001 -0.22	-0.099 *** 0.06 -5.13				

(a) Since the stock returns are already adjusted by size, the coefficient is not expected to be significant

PANEL B: ]	Future stock	returns and a	accrual con	nponents						
$Ret_{it+1} = $	$\beta_0 + \sum_{k=1}^n \beta_k$	$ACC_{it} + \sum_{j=1}^{m}$		$+\varepsilon_{it+1}$	,	tended accrual ontrol variable.	•	ts)		
	Intercept	(	(-)4COL	∆COA	(-)ANCOL	ΔΝϹΟΑ	∆LTI	(-)∆FINL	⊿STI	$R^2$
mean coef.	-0.042 -1.08	Controls yes	-0.313 * -7.22	** <sup>*</sup> -0.278 -6.64	*** <sup>•</sup> -0.182 -3.43	*** -0.194 ** -6.06	-0.126 * -4.41	-0.042 -1.26	-0.098 ** -6.77	** 0.07
Persistence	Order <sup>(a)</sup>		1	2	3	4	5	6	7	

The sample consists of 100,787 firm-year observations for 1976-2013. Accrual variables are winsorised to +1 and -1, other variables are winsorised to 1% and 99%. X denotes controls (size, market Beta, B/P, E/P and past returns). See Table A in the Appendix for the definitions of other variables. Standard errors clustered by firm and year using the Petersen (2009) approach. \*\*\* denotes statistical significance at the 1% level.

<sup>(a)</sup> indicates the persistence order of a particular accrual component obtained from the accrual persistence tests reported in Table 3.

Table 10	
Forecast errors and accruals portfolios	

	m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
Intercept				***-0.015 ***						-		
dec∆WC	-0.006 ***	-0.006 ***	-0.004	***-0.003 ***	*-0.003 *	**-0.003 *	**-0.003	***-0.002	***-0.001 **	**-0.001	-0.001	-0.001
	-6.12	-6.52	-3.44	-2.90	-3.13	-5.02	-4.83	-3.7	-2.54	-1.89	-1.44	-1.77
dec∆NCO	0.003 ***	0.002 ***	0.001	0.001	0	0.002	0.001	0.001	0.001	0	0	0
	3.06	2.50	0.97	0.72	0.56	1.62	1.1	1.18	1.81	0.95	1.17	1.15
dec∆FIN	0.010 ***	0.009 ***	0.009	*** 0.008 ***	* 0.008 *	*** 0.006 *	** 0.006	*** 0.005	*** 0.004 **	* 0.004 *	*** 0.004	*** 0.00
	7.53	7.54	6.79	7.51	7.10	6.35	6.61	7.06	6.11	6.13	5.92	5.18

#### PANEL A: Forecast errors and accrual components portfolios

$\textit{Ferror}_{\textit{s,it+1}} = \beta_0 + \beta_1 \textit{dec} \Delta \textit{COL}_{it} + \beta_2 \textit{dec} \Delta \textit{COA}_{it} + \beta_3 \textit{dec} \Delta \textit{NCOL}_{it} + \beta_4 \textit{dec} \Delta \textit{NCOA}_{it}$
$+ \beta_5 dec \Delta LTI_{it} + \beta_6 dec \Delta FINL_{it} + \beta_7 dec \Delta STI_{it} + \varepsilon_{it+1}$

m1	m2	m3	m4	m5	m6	m7	m8	m9	m10	m11	m12
-0.009 **	**-0.008 ***	*-0.008	***-0.009	***-0.008	***-0.006	***-0.007	***-0.006	***-0.005 **	**-0.004	***-0.004 *	***-0.003 ***
	0.0									0.00-	
											-3.75
-0.006 **	-5.76	-1.97	-1.2	-0.003	-0.004 -4.76	-2.87		-2.2	-0.55	-0.46	-0 -0.19
-0.009 ** -6.72	•*-0.008 *** -6.38	-0.008 -5.74	***-0.007 -5.01	0.007-*** -4.98-	***-0.005 -5.42	***-0.005 -4.41	***-0.004	***-0.003 ** -4.41	**-0.003 -3.83	***-0.003 * -3.75	-3.38
-0 -0 32	-0 -0.6	-0.002	-0.001	-0.001	-0 0.52			0	0	0	0 0.56
0.001	0.002 **	0.002	** 0.002	0.002	0	0	0	0	0	0	0
• 0.009 **	** 0.008 ***	* 0.007	*** 0.007	*** 0.007	*** 0.006	*** 0.006	*** 0.005	*** 0.004 **	** 0.004	*** 0.003 *	0.15 0.003 *** 2.72
			*** 0.004	*** 0.004	*** 0.003	*** 0.003	*** 0.003	*** 0.002 **		*** 0.002 *	
	-0.009 *** -0.012 *** -9.69 -0.006 *** -5.48 -0.009 *** -6.72 -0 -0.32 0.001 1.59 0.009 *** 3.87	-0.009 ***-0.008 ** -0.012 ***-0.012 ** -9.69 -9.45 -0.006 ***-0.006 ** -5.48 -5.76 -0.009 ***-0.008 ** -6.72 -6.38 -0 -0 -0.32 -0.6 0.001 0.002 ** 1.59 1.98 0.009 *** 0.008 ** 3.87 3.65 0.004 **	-0.009 ***-0.008 ***-0.008 -0.012 ***-0.012 ***-0.008 -9.69 -9.45 -4.04 -0.006 ***-0.006 ***-0.004 -5.48 -5.76 -1.97 -0.009 ***-0.008 ***-0.008 -6.72 -6.38 -5.74 -0 -0 -0.002 -0.32 -0.6 -0.83 0.001 0.002 ** 0.002 1.59 1.98 1.90 0.009 **** 0.008 **** 0.007 3.87 3.65 3.63 0.005 **** 0.004 **** 0.004	-0.009 ***-0.008 ***-0.008 ***-0.009 -0.012 ***-0.012 ***-0.008 ***-0.006 -9.69 -9.45 -4.04 -3.04 -0.006 ***-0.006 ***-0.004 ***-0.003 -5.48 -5.76 -1.97 -1.2 -0.009 ***-0.008 ***-0.008 ***-0.007 -6.72 -6.38 -5.74 -5.01 -0 -0 -0.002 -0.001 -0.32 -0.6 -0.83 -0.48 0.001 0.002 ** 0.002 ** 0.002 1.59 1.98 1.90 1.92 0.009 *** 0.008 *** 0.007 *** 0.007 3.87 3.65 3.63 4.74 0.005 *** 0.004 *** 0.004 *** 0.004	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 -9.69 -9.45 -4.04 -3.04 -2.93 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -5.48 -5.76 -1.97 -1.2 -1.15 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 -6.72 -6.38 -5.74 -5.01 -4.98 -0 -0 -0.002 -0.001 -0.001 -0.32 -0.6 -0.83 -0.48 -0.52 0.001 0.002 ** 0.002 ** 0.002 0.002 1.59 1.98 1.90 1.92 1.77 0.009 *** 0.008 *** 0.007 *** 0.007 *** 0.007 '3.87 3.65 3.63 4.74 4.17	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 ***-0.006 -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 ***-0.006 -9.69 -9.45 -4.04 -3.04 -2.93 -5.48 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -0.004 -5.48 -5.76 -1.97 -1.2 -1.15 -4.76 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 ***-0.005 -6.72 -6.38 -5.74 -5.01 -4.98 -5.42 -0 -0 -0.002 -0.001 -0.001 -0 -0.32 -0.6 -0.83 -0.48 -0.52 0.52 0.001 0.002 ** 0.002 ** 0.002 0.002 0 1.59 1.98 1.90 1.92 1.77 0.92 0.009 *** <sup>*</sup> 0.008 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.006 3.87 3.65 3.63 4.74 4.17 3.73 0.005 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.003	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 ***-0.006 ***-0.007 -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 ***-0.006 ***-0.005 -9.69 -9.45 -4.04 -3.04 -2.93 -5.48 -5.35 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -0.004 ***-0.003 -5.48 -5.76 -1.97 -1.2 -1.15 -4.76 -2.87 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 ***-0.005 ***-0.005 -6.72 -6.38 -5.74 -5.01 -4.98 -5.42 -4.41 -0 -0 -0.002 -0.001 -0.001 -0 0 -0.32 -0.6 -0.83 -0.48 -0.52 0.52 0.52 0.001 0.002 ** 0.002 ** 0.002 0.002 0 0 1.59 1.98 1.90 1.92 1.77 0.92 0.98 <b>*</b> 0.009 **** 0.008 **** 0.007 **** 0.007 **** 0.007 **** 0.006 **** 0.006 <b>*</b> 3.87 3.65 3.63 4.74 4.17 3.73 4.29 <b>*</b> 0.005 **** 0.004 **** 0.004 **** 0.004 **** 0.004 **** 0.003 **** 0.003	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 ***-0.006 ***-0.007 ***-0.006 -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 ***-0.006 ***-0.005 ***-0.004 -9.69 -9.45 -4.04 -3.04 -2.93 -5.48 -5.35 -5.03 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -0.004 ***-0.003 ***0.0018 -5.48 -5.76 -1.97 -1.2 -1.15 -4.76 -2.87 -2.58 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 ***-0.005 ***-0.005 ***-0.004 -6.72 -6.38 -5.74 -5.01 -4.98 -5.42 -4.41 -4.6 -0 -0 -0.002 -0.001 -0.001 -0 0 0 -0.32 -0.6 -0.83 -0.48 -0.52 0.52 0.52 0.61 0.001 0.002 ** 0.002 ** 0.002 0.002 0 0 0 1.59 1.98 1.90 1.92 1.77 0.92 0.98 0.92 <b>*</b> 0.009 **** 0.008 **** 0.007 **** 0.007 **** 0.006 **** 0.006 **** 0.005 *3.87 3.65 3.63 4.74 4.17 3.73 4.29 4.53 <b>*</b> 0.005 **** 0.004 **** 0.004 **** 0.004 **** 0.003 **** 0.003 **** 0.003	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 ***-0.006 ***-0.007 ***-0.006 ***-0.005 *** -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 ***-0.006 ***-0.005 ***-0.004 ***-0.003 *** -9.69 -9.45 -4.04 -3.04 -2.93 -5.48 -5.35 -5.03 -3.41 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -0.004 ***-0.003 ***0.0018 ** -0.002 ** -5.48 -5.76 -1.97 -1.2 -1.15 -4.76 -2.87 -2.58 -2.2 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 ***-0.005 ***-0.005 ***-0.004 ***-0.003 ** -6.72 -6.38 -5.74 -5.01 -4.98 -5.42 -4.41 -4.6 -4.41 -0 -0 -0.002 -0.001 -0.001 -0 0 0 0 -0.32 -0.6 -0.83 -0.48 -0.52 0.52 0.52 0.61 1.09 0.001 0.002 ** 0.002 ** 0.002 0.002 0 0 0 0 1.59 1.98 1.90 1.92 1.77 0.92 0.98 0.92 -0.02 <b>*</b> 0.009 *** <sup>*</sup> 0.008 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.006 *** <sup>*</sup> 0.006 *** <sup>*</sup> 0.005 *** <sup>*</sup> 0.004 *** <b>*</b> .0.004 *** <sup>*</sup> 0.008 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.007 *** <sup>*</sup> 0.006 *** <sup>*</sup> 0.006 *** <sup>*</sup> 0.005 *** <sup>*</sup> 0.004 *** <b>*</b> .0.005 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.004 *** <sup>*</sup> 0.003 *** <sup>*</sup> 0.003 *** <sup>*</sup> 0.003 *** <sup>*</sup> 0.003 *** <sup>*</sup> 0.002 ***	-0.009 *** - 0.008 *** - 0.008 *** - 0.009 *** - 0.008 *** - 0.006 *** - 0.007 *** - 0.006 *** - 0.005 *** - 0.004 *** - 0.002 *** - 0.002 *** - 0.002 *** - 0.002 *** - 0.002 *** - 0.006 *** - 0.003 *** - 0.003 *** - 0.002 *** - 0.006 *** - 0.006 *** - 0.006 *** - 0.003 *** - 0.003 *** - 0.002 *** - 0.006 *** - 0.006 *** - 0.004 *** - 0.003 *** - 0.003 *** - 0.002 *** - 0.006 *** - 0.006 *** - 0.004 *** - 0.003 - 0.003 - 0.004 *** - 0.003 *** - 0.002 *** - 0 - 5.48 - 5.76 - 1.97 - 1.2 - 1.15 - 4.76 - 2.87 - 2.58 - 2.2 - 0.55 - 0.009 *** - 0.008 *** - 0.007 *** - 0.007 *** - 0.005 *** - 0.004 *** - 0.003 *** - 0.003 - 6.72 - 6.38 - 5.74 - 5.01 - 4.98 - 5.42 - 4.41 - 4.6 - 4.41 - 3.83 - 0 - 0 - 0 - 0.002 - 0.001 - 0.001 - 0 0 0 0 0 0 - 0.32 - 0.6 - 0.83 - 0.48 - 0.52 0.52 0.52 0.52 0.61 1.09 0.45 - 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.004 *** 0.002 *** 0.002 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.003 *** 0.002 *** 0.004 *** 0.003 *** 0.003 *** 0.003 *** 0.002 *** 0.002 *** 0.002 *** 0.003 *** 0.002 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.003 *** 0.003 *** 0.003 *** 0.003 *** 0.002 *** 0.002 *** 0.002 *** 0.002 *** 0.003	-0.009 ***-0.008 ***-0.008 ***-0.009 ***-0.008 ***-0.006 ***-0.007 ***-0.006 ***-0.005 ***-0.004 ***-0.004 *** -0.012 ***-0.012 ***-0.008 ***-0.006 ***-0.006 ***-0.006 ***-0.005 ***-0.004 ***-0.003 ***-0.002 ***-0.002 ** -9.69 -9.45 -4.04 -3.04 -2.93 -5.48 -5.35 -5.03 -3.41 -3.93 -3.91 -0.006 ***-0.006 ***-0.004 ***-0.003 -0.003 -0.004 ***-0.003 ***0.0018 ** -0.002 *** -0 -0 -5.48 -5.76 -1.97 -1.2 -1.15 -4.76 -2.87 -2.58 -2.2 -0.55 -0.46 -0.009 ***-0.008 ***-0.008 ***-0.007 ***-0.007 ***-0.005 ***-0.005 ***-0.004 ***-0.003 ***-0.003 ***-0.003 -6.72 -6.38 -5.74 -5.01 -4.98 -5.42 -4.41 -4.6 -4.41 -3.83 -3.75 -0 -0 -0.002 -0.001 -0.001 -0 0 0 0 0 0 -0.32 -0.6 -0.83 -0.48 -0.52 0.52 0.52 0.61 1.09 0.45 0.23 0.001 0.002 ** 0.002 ** 0.002 0.002 0 0 0 0 0 0 1.59 1.98 1.90 1.92 1.77 0.92 0.98 0.92 -0.02 0.55 0.26 0.009 *** 0.008 *** 0.007 *** 0.007 *** 0.007 *** 0.006 *** 0.006 *** 0.005 *** 0.004 *** 0.004 *** 0.003 * '3.87 '3.65 '3.63 '4.74 '4.17 '3.73 '4.29 '4.53 '3.37 '3.38 '2.92 0.005 *** 0.004 *** 0.004 *** 0.004 *** 0.004 *** 0.003 *** 0.003 *** 0.003 *** 0.002 *** 0.002 *** 0.002

The number of firm-year observations ranges from 45,145 to 48,142 across twelwe months from 1976 to 2013 for which consensus analyst earning forecasts and actual earnings are available on the IBES summary statistics file. Standard errors clustered by firm and year using the Petersen (2009) approach (see section 6 for the definition of variables). Decile accrual portfolios are formed based on the magnitude of a particular accrual component in year t, and scaled to a (0,1) range. \*\*\*, and \*\* denote statistical significance at the 1% and 5% levels, respectively.